



Moyixi Lei ^{1,2,†}, Longqin Xu ^{1,†}, Tonglai Liu ¹, Shuangyin Liu ^{1,*} and Chuanheng Sun ^{2,*}

1

- College of Information Science and Technology, Zhongkai University of Agriculture and Engineering,
- Guangzhou 510225, China; leimoyixi@163.com (M.L.); xulongqin@126.com (L.X.); liutonglai@163.com (T.L.)
- ² National Engineering Research Center for Information Technology in Agriculture, Beijing 100097, China
- * Correspondence: liushuangyin@163.com (S.L.); sunch@nercita.org.cn (C.S.)

+ These authors contributed equally to this work.

Abstract: Concern about food safety has become a hot topic, and numerous researchers have come up with various effective solutions. To ensure the safety of food and avoid financial loss, it is important to improve the safety of food information in addition to the quality of food. Additionally, protecting the privacy and security of food can increase food harvests from a technological perspective, reduce industrial pollution, mitigate environmental impacts, and obtain healthier and safer food. Therefore, food traceability is one of the most effective methods available. Collecting and analyzing key information on food traceability, as well as related technology needs, can improve the efficiency of the traceability chain and provide important insights for managers. Technology solutions, such as the Internet of Things (IoT), Artificial Intelligence (AI), Privacy Preservation (PP), and Blockchain (BC), are proposed for food monitoring, traceability, and analysis of collected data, as well as intelligent decision-making, to support the selection of the best solution. However, research on the integration of these technologies is still lacking, especially in the integration of PP with food traceability. To this end, the study provides a systematic review of the use of PP technology in food traceability and identifies the security needs at each stage of food traceability in terms of data flow and technology. Then, the work related to food safety traceability is fully discussed, particularly with regard to the benefits of PP integration. Finally, current developments in the limitations of food traceability are discussed, and some possible suggestions for the adoption of integrated technologies are made.

Keywords: food safety; blockchain traceability; Internet of Things; artificial intelligence; privacy protection

1. Introduction

Ensuring food safety is a very complex task [1]. As food is not easily preserved and is highly mobile, information needs to be collected in real-time for monitoring and traced to individual supplier nodes [2]. However, these operations are often unsafe and inefficient [3]. Food is not easily transported, nor is it preserved from production to distribution, and it is prone to spoilage, which leads to waste [4]. Studies show that more than 30% of food is discarded directly because of spoilage of the food itself [5], and a third of food is wasted because of improper handling during transport [6].

One of the many reasons for these serious problems is the mismatch of information between upstream and downstream, and the inauthenticity of information uploaded and stored during the traceability process [7]. Although a small amount of information is available on the outer packaging of food, little is known about the processing, logistics, and distribution of the food [8]. This inefficient labeling information can reduce the self-confidence of suppliers in the traceability chain, give incorrect guidance to consumers, and even disrupt the ecological balance of the market. As a result, there has been an urgent need for food safety traceability.

Against this backdrop, several technologies are taking shape to protect food safety, and they offer specialist services for the rapid traceability and management of food products [9].



Citation: Lei, M.; Xu, L.; Liu, T.; Liu, S.; Sun, C. Integration of Privacy Protection and Blockchain-Based Food Safety Traceability: Potential and Challenges. *Foods* **2022**, *11*, 2262. https://doi.org/10.3390/ foods11152262

Academic Editors: Riccardo Testa, Giuseppina Migliore, Giorgio Schifani and József Tóth

Received: 4 July 2022 Accepted: 23 July 2022 Published: 28 July 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). For example, supply chains and cold chains can store food data, prevent data tampering, and provide food traceability queries [10,11]. Multi-chain technology can be used to store private information within a company during the food distribution process [12]. Cross-chain technology is suitable for collaborative product data compatibility and asset transfer between enterprises [13]. Yet these services themselves are highly dependent on traceability data, and the technology provided from the data itself is barely visible in the traceability process. Faced with such challenges, food traceability is a need for close cooperation between supplier nodes to provide more transparency in the traceability logistics process, such as in the transport and storage chain [14]. However, because of information barriers in food data, which is not precisely located and shared, it is highly susceptible to malicious tampering or destruction [15], resulting in intermediate links being isolated and then food being wasted. For consumers, curious about more than just the data on the label, this ambiguous information is highly likely to convey the wrong idea to them, at which point food safety will become a challenge facing the food traceability sector.

Food safety has come to occupy an important place in food traceability [16]. The emergence of the Internet of Things (IoT) technology marked a watershed between the first and second generations of traceability systems, automatically identifying and tracking food through marker codes and using wireless networks for real-time monitoring, promoting digitization and electronic development, with the US, China, the EU, and Japan announcing IoT initiatives one after another, and the era of traceability information beginning to develop rapidly. It was not until the emergence of Artificial Intelligence (AI), and its application in traceability systems [17] that the third generation of traceability was officially opened. Its superior computing power and optimized algorithms quickly adapted to the explosive growth of data [18]. The most recent leap forward was the introduction of Bitcoin in 2008 [19], and Blockchain (BC) became widely known as the underlying decentralized technology that combined the best of many disciplines, including computer technology, cryptography, and economics, to open up a new journey for food traceability systems.

These technologies can transform the efficiency of food safety by changing the way data is collected, processed, and stored in tracking systems, and they promise to generate greater demand for food tracking. Solutions corresponding to these technologies for food safety have been studied in the literature. However, there has been less research into the application of integrated privacy-preserving technologies to food tracking processes.

The objectives of this study are: (1) to review the research and technologies related to food traceability and to identify the main needs for food traceability; (2) to integrate IoT, AI, PP, and BC while analyzing the solutions and potential proposed in existing articles; (3) discuss the development gaps in blockchain traceability and gives suggestions for possible solutions. The focus of this study is on the integration of privacy protection technologies with food traceability processes. This work has informed future work in the area of food safety traceability, with the following key contributions.

- 1. Key data flows in the food traceability chain and the technical support covered.
- 2. A comprehensive review of security issues in food traceability integration technologies and solutions to meet the requirements of food traceability at each step.
- 3. Discusses how PP technologies can be integrated into the full food traceability process to meet all needs, presenting challenges and opportunities for solutions.

The rest of the paper is organized as follows. Section 2 introduces the concepts of food traceability (in the form of data flows) and the technical support and requirements considered in food safety traceability. Section 3 analyzes existing solutions in the kinds of literature on food safety traceability based on the five issues that need to be addressed in food safety traceability. Section 4 analyzes the possible reasons for the slow development of blockchain traceability technologies and the technical and theoretical gaps that exist in the integration of food traceability technologies. Finally, Section 5 concludes the paper and suggests future work.

2. Background of Food Safety

This section reviews the importance of food traceability through the processing of data flows for food traceability. Then it discusses the main technologies in the food traceability process considered in this study, and their potential applications in food tracing.

2.1. Food Data Tracing

The blockchain is a digital ledger of transactions maintained by a network of multiple computers that do not rely on each other [20]. The blockchain allows data to be processed, transmitted, stored and represented in a readable form of special software on the platform. In its original configuration, each block contains a header, an electronic stamp with time, transaction data, and a link to the previous block. A hash is generated based on the contents of each block leaf node, which is then inserted and referenced in the block header of the subsequent block. Each leaf node is a hash of the transaction data, and the SHA-256 (Secure Hash Algorithm 256) hash function is used in the blockchain to deal with transaction data of any length [21]. After the food source data is processed to obtain a string of 256-bit, the unified format of the characters is then packaged and stored in the block, which saves storage space and ensures data security at the same time (R1). The block is divided into the block header and the block body [22]; the two parts use the Merkle-Tree data structure for storage connection, as well as the leaf node transactions two by two to get the hash value, which is the leaf node of the previous layer; thus, the upward progression, layer by layer, allows all the transactions in the block body iteration to get the root hash value packaged and stored in both the block header and the packing transactions by adding key values such as transaction timestamps and random numbers. Therefore, any change to the block will cause the distributed data ledger on the chain to fail to complete consensus [23] (R1).

Blockchain is a distributed database technology developed based on digital cryptocurrencies [24]. Blockchain systems are decentralized, tamper-proof, highly autonomous, and distributed consensus, providing some solutions to the problem of achieving distributed consistency without third-party supervision (R2, R3). With the rapid development of smart agriculture, blockchain has become more popular in the application area of food traceability platforms, but it also faces the challenges of the imperfection of its system and privacy leakage. Blockchain traceability of physical state processes and data flow states in Figure 1. The data generated by the five key steps from output to purchase are correspondingly included in the five processes of data processing.



Figure 1. Blockchain traceability of physical state processes and data flow states.

Data collection: Most of the data collected on the farm are obtained from sensors, such as temperature sensors showing changes in temperature, light sensors showing changes in the duration of sunlight, and so on. It is also necessary to manually record information about the growth process of the agricultural products, important time points, processing processes, quality control departments, distributor manufacturers, retailers, and the like.

All information is selected in a logical and rational spatial distribution of nodes so that each node uploads its data on the chain according to the rules (R2).

Data storage: Data is stored in the mode of single-chain storage, or multi-chain storage [25] As the data redundancy of the blockchain itself is very large, there are as many full nodes as there are *N* times of redundant data, so it is impossible to store all the data itself, only the hash value of the data can be stored on the chain (R1). The risk to a database comes from external attacks, internal leakage, imperfect database system security, etc. External factors include Trojan virus implantation, database backdoor, and so on. Internal factors include insider theft, weak password configuration, database system vulnerability, there are SQL injection, XSS injection, and a series of attacks that can steal users' private data.

Data processing: The traceability data on the chain is considered to be a dataset [26]. In addition, the dataset includes a timestamp, a random number, version information, and a hash of the previous block (or a custom version) (R1). By using SHA2 calculations to obtain a string of hash characters, blocks form logical links to each other, and any change in a block affects the information in the previous and subsequent blocks of the current block. At the same time, the data written to the block is broadcast for validation and each vendor node's transaction information needs to be timestamped, quickly and accurately (R2). This is because the main causes of leakage threats can be summarized as improper classification and grading of data, human error, and leakage of the encryption process.

Data transfer: Due to the special way blockchain data is stored, the n consensus groups within it work in parallel, asynchronously, and independently. However, there is no better way to deal with data redundancy on the chain than reducing the size of the shards to improve the efficiency of traceability data transfers, which is both a strength and a weakness of distributed ledgers. The main transmission methods by which data is compromised are network hijacking and transmission congestion [27].

Data publication: Traceability data ultimately serve consumers [28]; they can get complete traceability information by scanning the traceability code, but the traceability process is transparent to users [29] (R2), so the publication of traceability data needs to go through strict examination and risk assessment before determining the content of information to be displayed.

2.2. Technologies in Food Traceability

Sustainable agriculture is the green hallmark of today's agricultural development, where food safety is a matter of concern for the health of every human life in the world [30]. Food safety has been divided into two categories, the edible safety of the food itself and the safety of the information associated with it. The food itself is safe for consumption because it has a shelf life [31]; whether it is naturally fresh food, such as seafood, fruit, and vegetables, or foods that require preservatives for preservation, they all have a best before date and need to be consumed before their shelf life expires. The addition of preservatives is harmful to humans and the waste of fresh food is not conducive to sustainable agriculture. However, blockchain, Internet of Things, artificial intelligence, and privacy protection allow for automated data collection and processing and are also being used to accommodate food safety (FS) requirements (Table 1). Based on the food data flows described above, this study considers the following major technologies involved in the traceability process (Table 2).

Table 1. Food safety requirements considered in this study.

Requirements	Description
R1	Unique and the uniform identification of food products.
R2	Transparent and authentic traceability data for food products.
R3	Secure and uniform protection of private data that is not publicly available.
R4	Real-time monitoring of food quality and safety during production, storage, and logistics.
R5	Decision support for resource allocation and product quality assessment.

Requirements		Techno	ologies	
	ΙοΤ	AI	РР	BC
R1		\checkmark		\checkmark
R2	\checkmark	\checkmark		\checkmark
R3			\checkmark	
R4		\checkmark		\checkmark
R5	\checkmark			
R6	-		\checkmark	\checkmark

Table 2. Technologies for requirements of food safety.

2.2.1. Internet of Things

The concept of the Internet of Things (IoT) was developed by the Massachusetts Institute of Technology (MIT) in the USA in 1999 [32]. Early IoT was used to describe an architecture that enabled physical transactions to share information from the natural world with users or other objects that could be controlled and acted upon remotely and automatically. The components that support this architecture are typically low-cost microcontrollers and embedded electronic chips equipped with sensors and wireless communication technologies.

Currently, there are 3 main types of sensors used in agriculture: physical property sensors, biosensors, and microelectromechanical sensors [33]. Physical property sensors achieve signal conversion through sensors that are sensitive to physical changes, such as thermistor sensors; biosensors use bio-sensitive elements to transmit information that responds to the biological world to physical or chemical transducers, such as cell sensors and microbial sensors; microelectromechanical sensors are the product of new technologies, which can be single or integrated, for precision processing or complex sensor networks [34]. In agricultural wireless sensor networks, the precise positioning of nodes is very important for the monitoring behaviors of the sensor network. Determining the location of nodes to identify where conflicts occur is important to prevent agricultural pests and diseases as well as to reduce the loss and waste of traceable food products. Wireless sensor networks (WSNs) are an important component of the IoT in agriculture. Data transmission technologies are widely used in the IoT in agriculture (Table 3). When transmitting agri-IoT information, communication technologies should be used selectively as the environment needs to change.

Table 3. Communication technologies of agricultural IoT.

Technologies	Application	Consumption	Transmission Distance	Advantage
Wireless WAN (GPRS/4G/5G/WiFi)	Voice, data	High	Long distance	Large coverage, high flexibility
Bluetooth (3.0/4.0/5.0)	Media, cable	Low	Within 10 m	Cheaper and simple configuration
ZigBee (1.0/2.0/3.0)	Monitoring, sensors	Low	Between 10 to 100 m	Low power, flexible network-topology
Lora	Data transparent transmission	Low	Long distance	Low power, stable operation

The ultimate aim of data processing is to collect and analyze the information collected. In the process of monitoring agricultural production, a large amount of production data is collected, making it real-time, dynamic, and large-scale. Using IoT, monitoring data can be stored and analyzed to a certain extent and uploaded to the next data processing node, for example in the food supply chain (R1, R4). One of the main uses of cloud computing technology is information processing, which can effectively address the storage, calculation, and associated processing of large-scale agricultural production data. Radio Frequency Identification (RFID) is the main technology of the Internet of Things [35]. It typically uses passive tags with a chip with a code and an antenna to transmit this information to a radio base. RFID usually requires a specific device with an antenna to read the information on the tag. However, in very humid, hot, and muggy environments, it is common for interference to occur during the RFID reading process [36]. As a result, in recent years, smartphones have been developed with NFC readers that can be controlled by an app [37].

These technologies can help identify product batches and be used for automated shipping and storage, avoiding input errors [38]. Single-board computers have a small form factor, high flexibility, ultra-ruggedness, and excellent performance [39] (R1). As such, they can monitor temperature, brightness, PH, or other physical units and use actuators to modify physical objects, such as turning on a sprinkler switch when a temperature set-point is reached. These devices can be installed in product production or traceability processes and can support real-time monitoring of the status of food and resources during the traceability process and how humidity and temperature regulation systems, such as greenhouse culture, are triggered (R4). The development of IoT technology, therefore, provides the basis for a complete supply chain traceability system.

2.2.2. Artificial Intelligence

One of the earliest approaches to artificial intelligence (AI) was through expert systems for speech recognition [40], which integrate expert knowledge of specific domains to create mechanisms for solving narrow problems; Ref. [41] has designed an agricultural voice information system based on a speech recognition system through customized keywords, improving complex touch-based operations. Simply verbalizing a simple word description is all that is needed to access it, and this also simplifies the system response process. In the process of rural informatization, a speech service system for agricultural information based on keyword speech recognition technology provides a convenient way to access agricultural information. Machine learning (ML) is currently the most popular artificial intelligence technique [42], aiming to improve the performance of its models by building learning behavior in machines through a software model that is trained over a large number of repetitions on sample data. It can often be used for prediction, classification, and clustering, and has applications in speech recognition, computer vision, and robot control, they can be applied at all stages of food production and supply. With the help of AI, the quality of traceable food can be optimally transported and stored, while the optimal transport route can be chosen to ensure its timeliness based on the environmental conditions of the transport or storage (R4, R5).

In agri-food, the application of AI can be generally divided into four stages: (1). Intelligent seed selection and testing (R4, R5). Seeds are directly related to the yield and quality of crops. Artificial intelligence in seed selection and testing can improve the purity and safety of seeds, and better the quality of food. (2). Intelligent soil irrigation (R4, R5). Real-time monitoring of soil moisture, with automatic and periodic irrigation to improve accuracy and water utilization; In reference [43], which uses cameras and computers instead of the human eye to identify, track, and measure targets for further image processing. With the development of computer vision, this technology has been widely used in the field of agricultural automation. Computer vision can be applied to the monitoring of healthy crop growth and can be used for the classification and quality inspection of agricultural products, such as the selection and testing of seeds to enhance their purity and safety for commercialization. In addition to the automated management of modern farms, unmanned farms are efficient and precise in their operations, saving manpower and material resources and the environment is fully protected. (3). Intelligent planting (R4, R5). The traditional agricultural planting process is time-consuming and labor-intensive, while the application of intelligent robots and robotic arms in all aspects of planting, management, picking, and sorting, fully realize intelligent and automated planting. In [44], it is mentioned that human-machine interaction, i.e., through robot control can be carried out to detect, grasp, sort, and transport fruits and vegetables. (4). Intelligent monitoring (R4, R5). Through big data analysis, can accurately predict the weather conditions and thus master the accuracy of watering. It is also possible to detect the growth of weeds and pest infestations. An

example of this is the work mentioned in [43] for the control of crop pests and diseases. Robots or robotic arms with camera equipment often referred to as visual servoing [45], have their own 'eyes' and use computer vision technology to perform task capture, making picking, irrigation, and application more precise. Thanks to the excellent computing power of artificial intelligence, the data can be processed in real-time and, through intelligent decision adjustment systems, the aim of optimizing the quality of the food can be achieved. For example, the image recognition [46], which maps small farms, uses the different spectra of the different crops in the farm as a benchmark and adjusts their clarity to divide the area, thus enabling precise application of medicine.

Finally, as AI is largely driven by big data, a large system of data collection is vital for AI. The Smart Information System (SIS) for agriculture is an integration of big data and artificial intelligence, where there are many different components from retrieval, analysis, and data processing. The types of data retrieved include biological activity, soil moisture, and microbial content, irrigation, climate, surface and crop temperatures, crop growth, and market pricing. Despite the potential value of applying agricultural big data in all these contexts, the volume of data currently being analyzed is still relatively small [47]. However, this still setting the stage for the country's rapid economic growth, as the effective use of big data in agriculture can bring enormous benefits. These include: improving water and air quality, reducing heavy metals in the soil, improving food quality and safety, preserving creatures diversity, and improving personal life quality, while increasing yields, reducing costs, making crop forecasts, and improving decision-making and efficiency [48] (R5).

Overall, artificial intelligence can cover the full spectrum from seed selection, cultivation, and crop monitoring to soil management, pest and disease control, and harvesting. The use of artificial intelligence in food security will not only improve yields but also enable green farming.

2.2.3. Privacy Protection

The processing of blockchain traceability data is divided into procedures as the collection, storage, use, processing, and transmission [49]. The publication of data, which can be put into use as a new output resource to generate economic benefits, and reuse can also promote economic and technological development. In blockchain food traceability, the data of the block has become an important production factor in the agri-food field. But the growing amount of data is getting massive day by day, and the type of data is getting more complicated with the increase of data volume, as well as the data requiring the blockchain system to be time-sensitive. However, if the data value is low-density, then the privacy data is not protected by extensive coverage; it is extremely easy to cause data leakage.

Privacy means that personal information must be protected and must not be exposed without explicit consent under any circumstances. Information about consumers and distributors can only be obtained if the privacy of food traceability is guaranteed. Trust is a product of the probability of an attack and the damage it can cause. Furthermore, trust involves two things: transparency and consistency. As it happens, transparency and consistency are the two most important concepts in food traceability systems. Based on these concepts, it is clear that privacy and security are key to gaining trust. Because malicious attacks will happen without protecting privacy and security. The list (Table 4) shows common attack types within IoT, AI, and BC. For example, an IoT service can gain the trust of its users and provide the relevant and appropriate security services, but it can still breach their privacy by revealing personal data without explicit permission [50].

Categories	Attack Types		
Data-based attacks	Data leakage, Unreal data injection attack, Database leakage,		
Data based attacks	Cryptographic-based attacks		
Network-based attacks	DoS/DDoS, Communication protocol attack, Side channel attack		
Application-based attacks	Phishing websites, Script viruses, DoS/DDoS		
	Third-party attacks, Software update attacks, Interception attacks,		
BC-based attacks	Replay attacks, state-fork attacks, Consensus-based attacks,		
	SC-based attacks		
I.T. has a distant	Sensor incidents, Sensor weakening, Untrusted node,		
101-based attacks	System hijacking		
	Virus attacks, Supply chain attacks, Man-in-the-middle attacks,		
other attacks	Social engineering analysis, Malware attacks		

Table 4. Attack types in the BC traceability.

In the supply chain model, different industry partners share common business interests but have different requirements and policies for protecting the privacy of sensitive and confidential business data and therefore some partners are reluctant to share their private and confidential data. To address these issues, allowed blockchains, such as Hyperledger, provide two private data sharing mechanisms, which are the query mechanism and the multi-channel mechanism [51].Therefore, 5 main data-based PP methods are discussed in this study as follows (Table 5).

Table 5. Privacy computing for information security.

Name	Description		
Differential privacy	Adding noise make it impossible for a malicious attacker to get one piece of real data out of many private messages.		
Federated learning	Combine multiple data sources to model and provide model inference and prediction services without local data from all parties.		
Secure multiparty computation	Techniques and systems for the secure calculation of agreed functions where participants do not share their data and where there are no trusted third parties.		
Homomorphic encryption	Data can be transferred, analyzed, and returned between different participants and the cloud without being viewed and in clear text.		
Zero-knowledge proof	The sender proves the authenticity of the data to the verifier, then completes the verification without revealing the real transmitted data.		

Differential privacy: Differential privacy (DP) protection technique is a popular technique for protecting the privacy of digital data [52]. The basic principle of DP is that the addition or removal of individual record tag numbers from data should not have a significant impact on the final output of a statistical query [53]. DP can Add noise making it impossible for a malicious attacker to get one piece of real data out of many private messages (R6). Therefore, the protection of an exact message can be accomplished by DP.

Federated learning: Federated learning (FL) is a distributed machine learning technique and system, it consists of two or more participants that perform joint machine learning through a secure algorithmic protocol that can combine multiple data sources to model and provide model inference, also and prediction services without the data of each party leaving the local area [54] (R5). In the FL framework, the participants only exchange intermediate computational results or transformations in cipher-text form and do not exchange data, to ensure that the data of each party is not exposed (R6). FL was first proposed by Google in 2016 and has quickly gained the attention of major internet companies, tech giants, and artificial intelligence companies. The FL application considered in this study focuses on information security for food traceability. In the [55], which presents a framework for PP data publishing in a FL environment (FedGP). The approach allows for the extraction of private, artificial data samples and the empirical assessment of the risk of information leakage (R5).

Secure multiparty computation: Secure multiparty computation (SMC) is a technique and system for securely computing an agreed function without the participants sharing their respective data and without a trusted third party [56]. Through secure algorithms and protocols, participants encrypt or transform data in plaintext form before making it available to other parties, so that none of the participants can access the plaintext data of the other parties, thus ensuring the security of each party's data (R6); In the [57], the author uses SMC for designing a PP language called Keep, which allows contracts to manage and use private data without exposing it to the public blockchain. In [58], a distributed block-tracking scheme based on secure multiparty computing is proposed that enables price comparisons to be made without revealing any individual user data to other users in order to protect privacy.

Homomorphic encryption: Homomorphic encryption (HE) allows the cipher-text to be processed and still be the result of encryption, maintaining homomorphism, and is, therefore, more concerned with the security of the processing of the data (R6). For example, in [59], HE can be used not only for blockchain data transmission security but also for the design of HE-based blockchain data security transmission methods by collecting asymmetric encryption public keys through IoT devices. HE is divided into partial homomorphic and full homomorphic, with partial homomorphic implementing only some specific operations, such as RSA. Full homomorphism, on the other hand, allows any operation that can be performed on encrypted data in plaintext without decrypting it, allowing deep and infinite analysis of encrypted information to still be performed without compromising its confidentiality. For example, By using a combination of BC and full HE, the limitations of information sharing in a super-connected supply chain can be remedied without compromising data control by using zero-trust information sharing [60].

Zero-knowledge proof: Zero-knowledge proof (ZKP) is a cryptography-based privacypreserving technique, essentially a protocol involving two or more parties (R6). The prover proves and convinces the verifier that it knows or possesses a message, but the proving process cannot reveal any information about the proven message to the verifier. There are two types of ZKP: interactive and non-interactive [61]. Interactive simply involves the verifier providing instructions to the prover and the prover giving an answer. For example, in [62], a new consensus mechanism for the quantum blockchain has been designed. This new consensus mechanism does not involve a large number of computing resources and is also resistant to a 51% attack. Non-interactive does not require communication between the two to complete the verification. For example, in [63], a multi-verifier zero-knowledge (MVZK) proof model is constructed to verify the truthfulness of each message round with extremely high scalability.

3. Literature Review and Discussion

To better understand how IoT, AI, PP, and BC can be integrated to meet the needs of food traceability, comprehensive integration of the literature treatment scheme has been described in this section. To pinpoint the selection of relevant literature articles, the study defined five research questions, as shown in Table 6. In the first step, we index papers according to many keywords, the keywords were selected in four gradations, firstly to determine the scope of application of the article, so "food" OR "agriculture" OR "agricultural" OR" agri-food" were chosen. Followed by the keywords "traceability chain" OR "supply chain" OR "traceability system", this was used to determine that the article only covered the area of food traceability. Further, the article was selected with "security" OR "safety" OR "differential privacy" OR "federated learning" OR "secure multiparty computation" OR "homomorphic encryption" OR "zero-knowledge proof" OR "privacy-preserving", these are only papers for food safety traceability. Finally, and most critically, select "Internet of thing" OR "RFID" OR "artificial intelligence" OR "expert systems" OR

"machine vision" OR "robot control" OR "blockchain" to identify existing food safety traceability articles containing only two or more these technologies.

Questions	Description		
Q1	How do PP and IoT help food traceability monitor storage status in real-time traceability?		
Q2	How can blockchain securely verify and share food data on the traceability chain?		
Q3	How does the combination of IoT and blockchain ensure the identification of traceable food and resources?		
Q4	How can IoT, PP, and AI ensure the secure storage of traceability data without tampering?		
Q5	How do AI and PP help traceability systems make effective and intelligent decisions?		

Papers for the literature review were selected primarily from the following publishers: Elsevier, ACM, Nature, Springer, Taylor & Francis, IEEEXplore, MDPI repository, and several websites. Therefore, the major index order is here: ("food" or "agriculture" OR "agricultural" OR "agri-food" and "traceability") AND ("traceability chain" OR "supply chain" OR "traceability system") AND ("safety " OR "security" OR "differential privacy" OR "federated learning" OR "Secure multiparty computation " OR "homomorphic encryption" OR "zero-knowledge proof" OR "privacy protection") AND ("Internet of thing " OR "RFID" OR "artificial intelligence" OR "expert systems" OR "machine vision " OR "robot control" OR "blockchain").

The technical requirements presented above (Table 1) will be discussed to answer each of the food traceability security questions: (1). IoT and identification privacy safety; (2). traceability; (3). food safety management; (4). traceability system management; and (5). decision support. The answers to the literature research questions will be answered under the R&Q discussions.

3.1. IoT Information Gathering

Interference with IoT privacy comes from two main sources, the limitations of external IoT devices, such as IEEE 802.15.4, and the effects of internal heterogeneity such as web servers and net clouds [64]. The former scholars have studied that network-based approaches are less vulnerable than host-based approaches, therefore, trade-offs because of the inability of static perimeter defenses to protect IoT devices, which are deployed inside the network and whose physical objects and computational data are constantly changing (Q1). The latter simplifies protocol design and system architecture in response to the complexity of the network.

Private data from sensors contain personal information about the participant, such as identity, location, or biometric data. For example, the identity of the operator is known through intelligent identification (Q5), location information is obtained from a GPS receiver, or location information on a Wi-Fi or cellular network [65]. In addition, a combination of temperature, light intensity, and orientation in the environment may also reveal location information. The leakage of location information may reveal participants' privacy, such as home address and workplace location, daily routines, and habits. Data from the source of privacy may contain sensitive information such as financial records, proprietary research data, or personal health information. In summary, the improvements in PP in IoT technologies are (1). authentication and authorization, (2). data anonymization and noise addition, and (3). data forgetting and data aggregation (Q1).

In addition, as IoT devices regularly collect usage status from users, they may contain sensitive information such as energy consumption or location information. Therefore, it is mainly used in the IoT environment for security protection of data sharing [66], and this type of approach can to some extent solve the trust crisis in the product supply chain (Q1, Q2).

Identification

A unique ID is an identity, and each participant on the BC traceability possesses to enable downstream nodes to obtain the flow of product batches from each link in the traceability chain [67]. Participants, such as containers, warehouses, or operators, also need a valid ID. There are many forms to identify a participant or a logical resource, but usually, a central authority is needed to maintain these ids. In BC, any proposal uploaded via the SDK is adopted and a unique ID number is assigned to that proposal, which is associated with the user's key (Q1). Because of the different nature of BC, their IDs function differently. In federated blockchain [68], the proposal signature is used to identify the number of user IDs (Q1). In some private blockchains that are extremely private, such as [69], a different signature is used to identify the validity of the user's identity (Q1). The widespread adoption of IoT facilitates its device usage and the data collection that can be stored for storage and analysis outside of the user's control domain.

Radio frequency identifiers (RFID) is widely used in IoT scenarios to identify things, record metadata, and control targets at a distance via radio waves. For example, [70], a framework for a block-based supply chain tracking system in radio frequency identification (RFID) technology has been developed(Q1, Q2). In the [71], the author states that vulnerable tags are subject to spying attacks and denial of service attacks. Unauthorized readers have the privilege of accessing these vulnerable tags without proper access rights; As described in the [72], which proposes a solution to control low-cost tagging schemes by implementing RFID access that can be well protected against open and unauthorized access to user privacy breaches (Q1).

Furthermore, in the [73], which also applies HE to the transfer of data from the IoT to cloud servers, and the algorithm acts as outsourced storage and computation for PP in the cloud-IoT model (Q1). Simulation results confirm that data transferred between IoT devices and the cloud is more private than existing methods. Moreover, as mentioned in the [74] which proposes a blockchain-based IoT combined with HE that can guarantee a high level of privacy for IoT data in a decentralized model (Q1). A quantitative framework was proposed to analyze the security of proof of work in blockchains, where the inputs to the framework include security, consensus, and network parameters [75] (Q2).

3.2. Traceability with Safety

The development of blockchain technology has extended to all areas of society, and also is becoming one of the most promising technologies with the potential to increase the transparency of communication, storage and transactions to even higher levels [76]. The application of blockchain to agricultural traceability is a completely new area of development, with blockchain enabling end-to-end traceability by bringing a common technical language to food products while allowing consumers to access the food information on their labels via their mobile phones (Q1, Q3).

For food products traceability, from the planting of crops to the final ability to be served and consumed, there are strict quality control steps and complex processing procedures [77], which include: seedling, planting, harvesting, processing, storage, cold chain logistics, quality control, sales, and consumption (Q2). As well as recording the impact of various parameters such as temperature, water quality, and soil elements (Q2). Inspection and quarantine information (heavy metal category and content, pesticide category and residue content, microbial categories and content, testing and inspection methods, quality level of agricultural products, inspectors, and inspection units) (Q2). Qualification management (enterprise qualification, management system, implementation standards). Other information such as plots (plot number, time, production batch, soil key indicators, etc.) (Q2). For example, in the [78], the author identifies the asset, threat, and countermeasure relationships associated with supply chain traceability, and the resulting hazard relationships can be addressed through PP techniques (Q2).

All the above information will be deposited in the distributed ledger of the blockchain as a depository, and a consensus will be constantly reached in the process of data verification (with a different timestamp) [79]. The data will be used as evidence of food safety proof, and any tampering with the information in any block will cause serious consequences. At the same time, the data added to the chain also helps retailers to determine the safe consumption date of foods. The changes in business models enabled by BC technology can lead to greater trust and transparency in the entire foods traceability ecosystem, and create new links to value exchange (Q2).

One of the biggest challenges in such a relationship, whether sold by individuals or run by businesses, processed by producers, or purchased and consumed by consumers is the security of BC traceability. Information in the traceability process, such as the amount of fertilizer used, the number of additives used, and the storage conditions should be available to all participants in the traceability chain. But doing so also increases the risk of data leakage. Therefore, in a traceability chain, individual links use identification codes and participants can know the exact location of the food product, temperature and humidity changes, and processing information (Q1). For instance, Provide secure and efficient data storage, data access, data sharing, and data authenticity services [80] (Q2).

As described in [81], there are smart contracts in food traceability to manage the triggering transactions of the logistics process (Q2). There are different response strategies for different trigger events. Facing consumers, the traceability chain provides traceability codes corresponding to each food product, so that there is no leakage and no duplication (Q2). In [82], a distributed recommendation system based on FL is used to model each service provider as a context-aware distributed online learner. Learning the user preferences, personalized recommendations are made based on the user context and historical behaviors. Perhaps this will help in the design of smart contracts and reduce the occurrence of errors and omissions (Q2, Q5).

In the reference [83], the authors ensure that product data collected from all stages of food traceability via sensors is legitimate while adhering to the terms agreed upon by the parties involved in the system, a cloud platform was developed where participants could access this data using a mobile application. However, there are confidence and security issues with a centralized solution. In a centralized repository, certain parties can further alter information to provide data that is not true for its purposes. Combining two technologies, IoT technology, and blockchain, improves data in real-time and ensures data security, for instance, and secure information transfer and storage are also better (Q1, Q2). In the reference [84], the authors propose an alternative approach that uses DP and SMC for the trade-offs of trust and safety (Q5). Combining DP with SMC allows for reducing the growth of noise injection as the number of parties increases without sacrificing privacy while maintaining a predefined trust rate.

3.3. Food Safety Management

With the help of blockchain technology developments, shared traceability information is guaranteed. In addition to this, the regulation of the quality and safety of agricultural products is regulated. The author Hua et al. standardized agricultural resource statistics and designed a system to record the flow of data at various stages of agricultural product traceability [85] (Q2). To require data transparency and build a blockchain supply chain model used to solve problems such as poor traceability communication was analyzed for blockchain traceability products [86] (Q2). Moreover, studies [87] have proposed privacy-preserving, scalable and efficient ways to ensure that IoT sensor data is collected, processed, and exchanged without tampering (Q1, Q2).

Both in [88,89] propose blockchain systems that are closely integrated with the IoT, with the former blockchain-based solution for agri-food supply chain management tracking providing a proposed API architecture that includes control transformation function calls

to the corresponding blockchain layer and the main system components of the blockchain itself components(Q1, Q2). In [90], the researchers propose a method for food data storage sources to logically control the production and supply chain of local agricultural products through smart contracts(Q2). Subsequently, in [91], the authors propose and design a "hash lock + smart contract + relay chain" framework to secure it across chains. In this way, it appears that there has been much research into the management of food security (Q2).

3.3.1. Food Products Trust

Trust is one of the most important factors in a traceability system [92]. Because people are increasingly curious about traceability information for agricultural products, they are also more concerned about whether the agricultural products they buy are within their shelf life. Whether it contains excessive amounts of preservatives. And as smartphones are becoming more common, people are more willing to pick up their phones and scan the QR codes on product packaging casings to verify their authenticity (Q1), which is a good thing for companies and consumers alike, as mentioned in the [93].

In [94], the authors suggest that, in China, the Grain Statistics Bureau does annual summary statistics on national grain production and harvest, which includes various types of information on agriculture, forestry, fisheries, and animal husbandry, along with the number of machines used for harvesting, year-on-year growth rates from previous years' production and harvest, the distribution of the agricultural products population and the number of imports and exports in the chart. With the formation of an agricultural traceability chain, more farmers are choosing to expand their industries by increasing their capital through agricultural insurance [95] (Q1, Q2, Q3, Q4).

In the case of agricultural insurance [96] and credit loans [97], again farmers are required to submit their bank information as a basis (Q3, Q4), as well as their own identity information (these are personal). A very large proportion of the data published each year is private (Third-party organizations simply process the data), but it is only a matter of time before privacy leaks cannot be stopped if an adversary conducts data analysis or traffic analysis [98]. In the [99], the authors focused on the use of blockchain for access control management or privacy security of identities, a decentralized personal data management system was created to address privacy leakage from users' use of third-party mobile platforms (Q2). The above studies focus on identity privacy because the blockchain is anonymous, but do not take into account PP for the original data resources. Hence, in [100], each IoT node was treated as a node in a blockchain to secure IoT nodes and use different privacy mechanisms to protect the privacy of each node's data (Q1, Q2).

In the same way, the margin for human intervention in blockchain-managed supply chains could be reduced significantly [101]. However, it must be considered that such phenomena have occurred in all previous technological revolutions, which have in turn demanded new skills and capacities in the labor market. Secondly, there are pests and diseases during the growth of agricultural products, and pesticides must be used to treat crops so that quality control is carried out on time (Q3). Furthermore, different agricultural products have different shelf life periods, such as seafood mostly only a few days, while vegetables can be stored for about a week and nuts for longer (Q2). The processing department's entry of the shelf life of food is something that needs to be strictly controlled, and if there is bad weather or an accident in transit, the time is shortened as much as possible. Distributors and retailers should finally label their goods with their unique labels, and all the above information should be verified to confirm that it is correct before putting the goods on the shelves for sale. For instance, in the [102], a restaurant prototype is proposed to implement more reliable food traceability using blockchain and product identifiers. The prototype has data from various stakeholders throughout the food supply chain. In addition to enhancing the traceability of food products, the prototype helps to grade the quality of food for human consumption (Q2).

3.3.2. Food Information Publication

Blockchain traceability technology is a fast-growing business that can respond quickly to needs across the country and the world, such as new crown virus traceability [103] and cold chain traceability [104]. Yet blockchain traceability is affected by a combination of environmental, political, and religious factors, which will be the biggest threat they face [105]. Blockchains are distributed databases (or ledgers) of encrypted records or digital events that can be shared between collaborations, with a secure data structure and redundant data collection allowing for random data validation (Q2). The project tracks the history of all steps included in the production, processing, and distribution system of agricultural products, which includes tracking the produce from its origin to the shelf for sale (Q1).

A traceability system using blockchain can reduce the traceability process from two years to two seconds, simplifying the process in between and making the entire process of traceability transparent [106]. Because adulteration is an important problem, consumers are now increasingly concerned about the raw materials and ingredients of agricultural products, and it is important to identify the causes of agricultural products born illnesses at an early stage, to find the source of agricultural products contamination, and to come up with reasonable solutions [107] (Q1). Likewise, the ingredients and processes of agricultural products should be clearly labeled to prevent specific human beings from developing life-threatening allergies to certain ingredients. There is also a need to clearly label the processing of agricultural products for different religions and ethnicities. This information is critical and randomized using privacy mechanisms that can be considered to protect the privacy of the consumer and the interests of the producer.

The food economy is gradually moving from offline to online and agricultural ecommerce is beginning to take off. The development of agricultural e-commerce platforms has an income-generating effect on food products, and the vigorous development of agricultural e-commerce can promote national economic development and strengthen the economy [108]. As descripted in the [109], the authors investigate the construction model of an e-commerce agricultural products online marketing system based on the blockchain end and an improved genetic algorithm (Q2, Q5). It automatically divides transaction profits through smart contracts, improves execution efficiency, and reduces transaction costs. Ultimately establishing a transparent, efficient, and applicable blockchain architecture for food trading.

3.4. Traceability System Management

The advances in security should not be underestimated, and more produce traceability companies are choosing to use security outsourcing products [110], while this mitigates the leakage and poor protection of blockchain data in some areas (Q2). This introduces a new threat that could result in a breach attack [111] or the implantation of a virus [112], which is extremely costly from a business perspective, not only in terms of internal losses but also in terms of being sued for poorly protecting the privacy and security of user data. The design as follows is that each layer of the blockchain architecture has different measures for PP, making the blockchain technology rigorous enough.

Data Layer The data layer is the core area of blockchain technology, this layer contains the digital signatures, timestamps, random numbers, and numerous transaction data of the data uploaded by each node. The data layer uses a tree structure to convert the transaction data into a hash string stored in the block body, using Md5 to take a digest. The method is irreversible in its calculation process and is protective of the transaction data, but not immune to tampering [113]. For example, the adversary gets the input by matching out the output of the hash, and although this is very computationally intensive, such an operation is sufficient to destroy the data structure in the block body.

From the point of view of PP, the randomness generated by DP can be considered a key aspect of protecting the privacy of hash functions from a PP perspective, e.g., SHA-256, SHA-2, SHA-1, etc. (Q2). In the [114], the user's public and private keys are protected

by cryptographic signatures, keys can be managed using PKI trust systems or IBC trust systems. Similarly, the integration of DP can help increase the privacy of some of the core components of the layer, or delegate HE to process data so that third parties cannot obtain any information about the plaintext from the cipher-text while maintaining the user's ownership of the data.

Network Layer In the blockchain, the network layer plays an important role in the timely decentralization and dissemination of messages through effective communication protocols [115]. The aim is too inter-transmit messages between authorized nodes by implementing the data from the data layer above, and data messages can be of different types [116]. For example, in Hyperledger, it is the authorized nodes that verify the broadcast data, but in Bitcoin, it is the consensus verification of the data by all nodes. Therefore, network congestion is a deadly killer of communication protocols, and good communication efficiency is required to ensure smooth consensus.

Various solutions have been proposed for security at the network layer by many scholars who use various techniques to improve and enhance communication efficiency and security, such as generative adversarial networks [117], Bayesian training networks [118], and the 6G wireless communication technology [119]. Without efficient PP in the communication protocols, adversaries can still easily eavesdrop on networks, attack devices, and obtain data [120]. However, protecting the private information of senders and receivers by adding randomness to propagation and broadcast, and protecting cross-network addresses by adding some noise before sending messages can ensure private information in P2P networks (Q1, Q2). Another method is to re-encrypt the proxy, switching keys midway through the process to ensure the security and integrity of the data, or simply use the HTTPS encryption protocol with SSL/TLS, which also prevents data leakage during data transmission (Q1, Q2, Q3).

Consensus Layer The consensus mechanism of the consensus layer, which can contain one or more, is one of the core parts of the distributed ledger and aims to decentralize the process of validating different data between nodes against each other to reach agreement [121]. The consensus algorithm varies by various blockchain networks and types, for example, Hyperledger Fabric, used in traceability systems, uses Kafka to consistently sort transaction data awaiting endorsement validation, while using the Practical Byzantine Fault Tolerance (PBFT) algorithm to increase the fault tolerance of the system (Q2). With the different integration requirements of different platforms, a large number of variants of consensus mechanisms have been developed, such as PoS [122], DPoS [123], PoA [124], etc. Given the extensive computational nature of consensus, edge computing can be used to refine the workload of the centralized server-like approach to clients at the edge, which reduces the loss incurred by the distributed computing and allocates computing power to other resources, and AI can enhance the availability of data (Q5).

Contract Layer Smart contracts are great for a blockchain network and determine the flexibility of that network. Various types of scripts, code, smart contracts, and algorithms are used and deployed in this layer, enabling complex transactions and functions to be integrated and work in a decentralized blockchain network. Again, different systems in this layer use different scripts, a base traceability model as Figure 2. Conditions are automatically triggered during the transaction to complete the data calculation, communication, and consensus process, without human intervention in the system process (Q1, Q2). From a security perspective, some researchers have counted existing smart contracts to protect data security, such as Shadoweth [125], Arbitrum [126], Raziel [127], and so on. Some of these works use cryptographic mechanisms to protect privacy, some use anonymization mechanisms, and some use privacy-enhancing mechanisms for the generation of public and private keys (Q2).



Figure 2. The internal structure of the safety traceability system.

Application Layer Because all data is presented directly to the user through the API connection interface, the application layer is also the most vulnerable to hacking. Hackers attack the backend or disrupt the availability of the entire interface by entering commands into the interface's input boxes, which directly affects the performance and network structure of the entire blockchain system. For example, the DAO attack directly led to the theft of at least 50 million dollars of Ether tokens "Ether " was stolen, necessitating a hard fork of Ether to protect the remaining funds [128]. PP plays an important role in providing PP to the blockchain application layer by ensuring the randomness of the network (Q2). For example, DP can reduce the risk of attacks by providing random data on the application layer, rather than providing all accurate data about other transactions and users. Therefore, DP plays a key role when an adversary has access to a set of transactions or data, but cannot confidently tell if a particular user data exists (Q2).

Similarly, can be protected data by randomizing the evaluation of any type of query performed on public blockchain data (Q2). For example, if an adversary gets the system output data through regular input, but there is no correlation or pattern between the output data, the degree of PP of the data is positively correlated with the keywords entered into the search, indicating that the adversary would have no way to get any information about the data other than his query. As mentioned in the [129], the authors propose a blockchain-based solution for supply chain data sharing in an information IoT environment. Attribute-based cryptography is used to manage access control of data in the IoT (Q1, Q3).

A traceability system model incorporating PP technology (Figure 2). This model collects real-time data via IoT devices and all logistics and storage data are uploaded to the food traceability chain (Q2, Q3). This is an ideal model for traceability combined with PP. In addition, several researchers also proposed the BC model for food safety, and there are commonalities in the selection and use of systems in their papers (Table 7). Those that

choose Ethereum as their test platform are mainly applicable to the automatic triggering and pluggable nature of smart contracts, for example paper from [130,131]. As such, Ethereum is more suitable for using a single transaction structure, whereas Hyperledger Sawtooth allows for the definition of custom transaction structures, so the majority of Hyperledger Sawtooth-based experiments will choose to perform custom blockchains, for example paper from [132,133]. Most of the platforms that choose to use the Hyperledger Fabric are either federated or public chains, and they will also use smart contracts to automate the mission trigging, for example, the papers from [134,135].

Table 7. Related food traceability platforms.

Platforms	Paper Working & Ref.	
Ethereum	Fully decentralized AgriBlockIoT, a solution for agri-food supply chain man Agement [88]	
	A food safety traceability system based on the blockchain and the EPC Information Services using enterprise-level smart contract to prevent data tampering and sensitive information disclosure [130]	
	A permissioned blockchain structure that protects privacy, using group signatures and broadcast encryption for privacy protection of data, and a PBFT for network-wide verification [136]	
	An agricultural supply chain system model ensures security and trust and simplifies transactions and administrative procedures [137]	
	A transparent, reliable, and tamper-proof framework for the food supply chain using smart contracts to develop [138]	
	Double systems (Hyperledger Sawtooth & Ethereum) to develop dairy supply chain [139]	
	Data is stored in the Interstellar File Storage System (IPFS), then returns a hash of the data store on the blockchain which ensures a secure solution [140]	
	A comprehensive solution for the agri-food supply chain. Ensures delivery mechanisms for the agri-food industry supply chain (IPFS stored data) [141]	
	A customized smart contract semi-automatic system designed for the agri-food industry sector (honey as an example) [131]	
	FoodTrail deepens the advantages in terms of access, security, and accuracy [132]	
	Be used to provide greater asset traceability in today's food supply chains [142]	
Hyperledger Sawtooth	The proposed system allows consumers to reconstruct the history of a product down to its origin for verification of health and quality with a simple QR code scan [143]	
	An agri-food blockchain technology in Malaysia (pepper), especially extends transparency and traceability in permissioned blockchain [144]	
	The goal is to track egg products from farm to fork using blockchain and internet of things (IoT) enabled technologies [145]	
	Focus on assessing the impact that a blockchain traceability system may have on constrained sensing devices. (test on Ethereum and Hyperledger Sawtooth) [146]	
	The concept of using IoT devices in combination with a cold chain (Hyperledger Sawtooth) system [147]	
	A traceability system, introducing product code and timestamps to protect privacy, as well as using asymmetric encryption algorithms for verification and signatures [133]	

Platforms	Paper Working & Ref.		
	Customizes the existing Hyperledger architecture and adds a PP module based on DP to its smart contract [134]		
	Combining ZKP technology to ensure privacy and traceability in the food supply chain [148]		
Hyperledger Fabric	A decentralized application (DApp) to verify food quality and the agricultural supply chain because of its high involvement and transparency [149]		
	A platform ensures data availability and traceability. The identification of unsafe food can be prevented from entering and enhances food safety and reduces manual errors [150]		
	The system ensures the uniqueness of food products, and the authenticity and reliability of the blockchain source data are ensured through IoT [151]		
	A platform that combats information asymmetry and collusive relationships and provides a dual mechanism for validating crowd-sensing data [152]		
	Help the development of the supply chain industry and the refinement of other blockchain systems, striking a balance between privacy protection and security and public blockchain [135]		
	A framework for a supply chain traceability system based on Hyperledger Fabric and passive RFID [70]		

 Table 7. Cont.

3.5. Decision Support

Agricultural production requires large amounts of inputs, such as water, fertilizers, pesticides, energy, and labor to maximize the quantity and quality of food. Agricultural production is complex and diverse in terms of extreme environmental conditions, crop traits, and management strategies. In addition to data collection, environmental control, and traceability, data collected during the food production phase and the logistics phase are processed using big data analysis methods to develop important information for intelligent decision-making (Q3). As a result, the efficiency of logistics operations is improved. Based on environmental information about timestamps, shelf-life predictions [153] or blockchain-based smart contract triggers can be developed to avoid food waste and loss, and provide better quality products (Q5). Known how important big data is for model learning in artificial intelligence. In the [154], the author outlines the importance of big data, security, and privacy in big data and the various challenges that need to be overcome to apply machine learning techniques to big data (Q4, Q5).

Factors that contribute to food waste include both explicit and implicit factors. Intrinsic factors relate to the characteristics of the product organisms, such as type of pH, natural microbiota, and microbial counts. Extrinsic factors are changes in external ecological conditions. The implicit factors include contamination with heavy metals, internal cellular variation, and other conditions. Some of these factors can be monitored by the various sensors mentioned in the IoT (Q1, Q2, Q5). As mentioned in the [155] the Big Data IoT framework from was used in a weather data analysis situation. The study implemented clustering and sensor anomaly detection using a publicly available dataset (Q4). The framework has the potential to extract meaningful information from a relatively complex dataset.

Data analysis and data mining techniques can be used to analyze the data and predict growth trends and progeny reproduction of the food products obtained, which is a good approach for obtaining high-quality food products [156] (Q5). In this era of machine learning-based artificial intelligence, privacy has become a big issue. It is worth noting that PP issues in the context of machine learning are very different from traditional data PP, as machine learning can be both a friend and an enemy (Q5). In the process of data traceability, the feedback of real-time monitoring data is powerful in assuring the quality of food (Q1). At the same time, having IoT combined with AI technology allows for optimal path selection to reduce food waste and loss (Q4).

As mentioned in the [157], the authors detail a new framework for PP deep learning and discusses its assets. The framework places emphasis on ownership and secure handling of data and introduces a representation based on chains of commands. This scheme allows one to implement complex PP structures (Q5). Then, the information collected during monitoring can be used to simulate environmental links, weather, and road conditions in their natural state (Q4). It might also be possible to create and apply a simulated result to make better routing decisions (Q5). In [158], a blockchain-based fish farming and traceability platform (FBCT), IoT is applied to detect water quality and upload data to the traceability platform to rationalize and manage the traceability system of the fishery (Q2, Q3).

AI and data-driven computing methods are key to connecting the field of traceability research [159]. AI techniques have powerful information analysis capabilities in controlling irrigation, identifying pests and diseases, and harvesting crops (Q4). The goal of this research area aims to make intelligent decisions about the process of transport and storage of food traceability (Q5). Agricultural AI will build these solutions using knowledge-driven and human-in-the-loop learning paradigms aimed at dealing with food system diversity and biological complexity, efficiently capturing and utilizing food system data, and gaining user trust through interpretability, security, privacy, and fairness (Q2, Q4, Q5). Using technologies such as machine vision, image reproduction, and cognition, artificial intelligence can make accurate judgments and predictions based on the agricultural information obtained, leading to intelligent decision-making (Q5).

4. Challenges

Different approaches have been taken to experiment with food traceability. However, a gap has been noted between these solutions and their practical application. This is particularly true in developing countries. Most of the technologies used require more indepth research, integration with other technologies, and adoption in current food tracking operations. Some challenges still need further discussion.

4.1. Gaps in Development

Taking a broader view, broken down by region, shows the US has the highest number of blockchain funding rounds per quarter, with a whopping 43% in Q1 2022, followed by Asia and Europe (Figure 3). Of these, total funding for blockchain startups in the US amounted to \$58 billion, largely unchanged from the previous quarter. Meanwhile, New York again ranked first with over \$2.1 billion in funding in the blockchain space, with more than half of the funds going to Fireblocks, ConsenSys, and OpenSea. Silicon Valley ranked second with \$1.6 billion in funding. Blockchain funding across Asia reached \$1.5 billion in the first quarter, a 275% increase year-on-year. However, 85% of these portfolio companies are currently in early-stage development.

Table 8, lists the top 10 companies related to the global blockchain space in the first quarter, in order of funding amount, with only one company on the list in both China and the Bahamas, and the remaining eight in the US. Their fields can be expanded to include finance, gaming, healthcare, and bioscience, but unfortunately, there are essentially no agriculture-related results. Different approaches have been taken to experiment with food traceability, but a gap between these solutions and their practical application has been noted, especially in developing countries. Most of the technologies used require more in-depth research, integration with other technologies, and adoption in current food traceability operations.



Figure 3. Number of blockchain-related business funding rounds worldwide.

Table 8. Global top equity deals in Q1'22.

Company Name	Round Amount	Country
Fireblocks	\$550 M	USA
ConsenSys	\$450 M	USA
Yuga Labs	\$450 M	USA
FTX	\$400 M	Bahamas
Animoca Brands	\$359 M	Hong Kong
OpenSea	\$300 M	USA
Blockdaemon	\$207 M	USA
The Graph	\$205 M	USA
Alchemy	\$200 M	USA
Aleo	\$200 M	USA

4.2. Technical Challenges

Some research surveys show that, that the structure of agricultural links is not tight [160], while from the perspective of companies, their branding needs to be strengthened, and there is not enough security for products grown by farmers, which should be supported by the industry. This is also a problem faced by many developing countries, which are undoubtedly unable to meet the needs of their people in terms of both technology and services [161].

First, blockchain latency has always existed, with sequences of waiting for consensus to grow as throughput increases, taking anywhere from a few minutes to a few hours to complete validation of things until all participants have completed their confirmation of the ledger and the transaction is considered successful [162]. However, such an operation is extremely inflexible and is very detrimental to the development and progress of blockchain traceability. Most of the current projects are in developed countries, such as IBM, and Fabric, but no significant questions are raised around this in their conclusion. To this end, considering the cost, block propagation, and validation time in blockchain networks constrain TPS performance, and since many solutions to increase TPS comes at the cost of reduced security, bodyless block propagation (BBP) with a pre-validation mechanism can

increase block propagation speed without compromising security compared to protocols on Ethereum [163]. As a result, TPS will no longer be constrained by block propagation. Another way can also combine security and transmission, for example, Ref. [164] designed a smart contract integration technology off-chain and on-chain, using smart contracts to guarantee privacy isolated from third parties, using existing cryptography to provide digital signatures and authentication. But to ensure the viability of data regulation, auditing of data on the chain needs to be accompanied by its security. For example, Ref. [165] encrypted storage of records, use of blockchain to store data, auditing of data stored on the blockchain.

Second, for the detection of agricultural products relying on smart IoT sensors, a failure of a sensor will result in missing a certain part of important data, while sensors require a working environment that is as safe and trustworthy as possible and untampered with. Furthermore, many producers are self-employed and it is expensive for farmers to purchase, for example, on-site sensors, drones, and precision spraying equipment. This is a major difficulty for such smallholder economies. Furthermore, because of the multimedia, IoT faces security, authentication, and privacy issues, a theoretical framework with security and blockchain as the main enablers have been proposed [166]. Furthermore, future research studies can focus on using blockchain-enabled models to replace ERP systems, reduce food waste and improve supply management between stages in the food supply chain [142].

Many supply chains feature complex processes that involve combinations of different products, traceable units, or specific objects that can be traced, which are important to define within food traceability. At the same time, to improve the communication efficiency and security of data, some researchers have proposed more efficient 6G communication technology, by using blockchain to improve the communication efficiency and security of data and to allow secure access to smart networks, cross-chain data sharing, and other operations [119]. For example, in healthcare, a large number of researchers have identified several issues of privacy, security, interoperability, and reliability in the preservation of electronic records of patients, and have used blockchain technology to add a layer of protection and proposed several data protection frameworks [167], but from a security technology perspective, analysis has shown that the personalized attributes in electronic medical records can be tampered with by attackers or accessed by unauthorized users for malicious reasons [168]. These records should be coupled with the flexible randomness of DP to obtain output data that better protects the database security of both physicians and patients.

Third, the combination of blockchain and artificial intelligence is an inevitable product of a pluralistic world. The world's first blockchain-based artificial intelligence operating system, the AICP network, is officially launched, which can open up unlimited intelligence in a trusted ecology [169]. AICP can be applied in multiple fields. The AICP can be used in a variety of fields, including smart manufacturing, industrial robotics, smart healthcare, smart retail, mobile internet, smart finance, etc.

Artificial intelligence chip will provide an open-source platform with rich development tools to meet the development needs of the AI wave, and facilitate developers to develop AI DAPPs with a better user experience by combining them with actual application scenarios [170]. Furthermore, building a cross-industry and multinational blockchain alliance chain to integrate the resources and research results of all parties, provide a new set of value Internet for the original solutions [171]. In their industries, and promote the rapid development of blockchain in various industries. In particular, the development of agriculture, which has obvious advantages over other fields, will take a qualitative leap forward.

One may be an expert in one area, but one knows very little about the application of integrated technology [172]. Thus, all the technical problems mentioned above have one thing in common, the absence of a systematic system for training technical personnel. The lack of talent is a major issue facing the blockchain industry. For example, China, the largest country among developing countries, currently has over 120,000 blockchain companies, with over 40,000 new companies in 2021 alone, an increase of 55.1% compared

to 2020 [173]. In stark contrast to the rapid expansion of the industry, there is always a shortage of relevant talents in the market at present. There are only 67,900 technically skilled talents in China with the ability to promote the application of blockchain technology, products, and services on the ground, and most of them are transferred from the Internet, software, IT services, and other fields [174]. The contradiction between the supply and demand of talents is outstanding. Given the dilemma of a long and relatively difficult talent training cycle, it is urgent to build a vocational-technical skills training system.

4.3. Sustainability and Transparency

Waste is a major problem for food safety; agricultural waste can be caused in several ways, including loss of food and resources because of improper storage conditions throughout the supply chain, and unsafe and unhygienic practices in food processing. So, these will make the storage of information inaccurate at each step and waste both the environment and the producers. In addition, the over-processing, fine packaging, coloring, and coding of food is itself a waste [175]. Therefore, increasing the transparency of blockchain traceability systems is a powerful means of reducing waste [176]. It was then suggested that the use of blockchain technology for agricultural by-products and final waste use would contribute to sustainable development [177].

Transparency is the most crucial strategic step in building consumer trust and, in turn, transparency in the traceability chain can improve social sustainability and traceability. The stagnation of agricultural products often leads to food waste, which is the main problem with blockchain traceability systems [178]. Much of the waste in developing countries is during harvesting and processing, as some countries do not have more advanced operations to harvest and process produce promptly, and most produce is often inedible before it is released for sale because of inclement weather and human factors. To reduce food waste after purchase, consumers need food to have a longer shelf life, which can be achieved in part by reducing the time it takes for food to leave the farm. For example, it takes only a few minutes from customs to shelf sale, but it takes three days [179]. Smart inspection devices can also be used to more accurately locate and track information about the stability of produce, reducing unnecessary waste during transport. However, increased or reduced transparency effectively reduces these problems and when consumers are curious about traceability information, they can feel more confident about eating the food.

To achieve social sustainability, the agricultural supply chain must not cause consumers to worry about their food. Improving transparency and traceability in the food supply chain of information should increase social sustainability. Therefore, the first issue to consider is the prohibition of unnecessary waste, as the demand to meet the growing global population will become a burden on available resources shortly [180]. The second is to reduce the irreversible harm to the environment caused by the excessive consumption of resources to the extent of the blockchain because the planet is the home on which all depend as human beings and every generation has the duty and responsibility to protect it. Sustainable agricultural product traceability also needs to ensure profitability for all stakeholders, which means that the costs associated with newly incorporated technologies, such as IoT, AI, PP, and BC, or losses that occurred to wasted food, do not limit profitability.

4.4. Governance and Regulation

The UN's mandate to target sustainable development is conducive to improving global rural health, hunger, poverty issues, environmental conditions, and illiteracy rates [181]. At the same time, the USDA states that if one wants to have safe and healthy food in the future, then one needs to worry about climate impact and future sustainability [182]. Because of technological advances, the way of life of people around the world has improved considerably. Relatively more attention is paid to the development of urban areas than rural areas, but the sustainability of a country depends on the development of its rural areas, especially in developing countries [183]. Blockchain traceability is one of the most high-profile technologies, and countless integrated technical and theoretical models, projects,

and frameworks have been proposed and implemented to help overcome the various issues and challenges that farmers face in their daily lives. However, blockchain traceability sustainability needs to take into account social, environmental, and economic factors.

Some research scholars have found that the sensitivity of blockchain data stored after the propagation of multiple blockchains can pose new challenges for block information management [130]. On the one hand, permanent data visibility may eventually compromise privacy concerns, so large companies prefer to implement private and permissioned blockchains, which may bolster corporate monopolies. Conversely, blockchain has also been described as a potential authentication technology for some workers and organizations [184,185]. The increased automation of transactions and processes throughout the traceability system and the elimination of transactional intermediaries may significantly reduce human intervention, resulting in a phenomenon where information is no longer useful.

On the other hand, it is also important in terms of governance—for example, the need for a more rational representation of the prices set for traceable agricultural products (blockchain traceability of agricultural products also implies an increase in their prices). Hence, they are often the focus of product fraudsters. Policy regulation resides on both sides of the blockchain traceability system, it can be both a help and a hindrance [186]. For instance, real-time regulation of opioids controls human intake at the source and reduces the number of deaths because of overdose, but fundamental changes are needed in the collection and use of monitoring data linked to the implementation of effective services, treatments, and prevention methods [187]. However, when processing database information, priority should be given to improving the timeliness of the data, the representativeness of the sample, the linkage of the database, and increasing the flexibility to adapt to changes in the environment while protecting privacy of the survey participants. After analysis, DP can be integrated using a blockchain system to increase the security and usability of the data.

In terms of market analysis, there are already new means of threatening privacy taking place, and even though there is normative technical support provided by various countries beforehand for compliance in the process of generating, distributing, and translating QR codes, there is still a danger that QR codes can be replaced and maliciously tampered with. Therefore, ensuring food information security is also a challenge for food safety traceability. It is first necessary to ensure that the QR code of the data scanned by the user is correct and secure, as there is a possibility of the QR code being replaced or tampered with, which can put the user in a difficult position of privacy leakage [188].

5. Conclusions

This study provides an overview of the characteristics and applications of the IoT, AI, PP, and BC. A detailed analysis of information flows in food safety is provided. In the course of the literature review, several IoT and AI solutions related to the identification of logistical processes in traceable food were identified. These solutions work to reduce food waste and improve the efficiency of food traceability. PP algorithms guarantee the data security that is being overlooked with a full range of protection measures to improve the authenticity and security of the data, and blockchain is adapted to the transparency and trust needed for food traceability to protect the security and privacy of transactions.

The sustainability of the ecosystem is crucial in large-scale solutions, such as the cost and waste of identifiers. Blockchain platforms, consumption of storage compartments, and inefficient user registration. Intelligent decision-making consumes a large number of samples, etc. The combination of IoT and AI may be feasible, just like 5G and in-car networks, and it promises to lead the rise of logistics research. Government attitudes and systems will be the restraining force in the development of various technologies for food traceability, because of differences in national conditions and religions. Monopolies are growing, but medium and small businesses are being weakened. Proposals and evaluations of new architectures for integrating PP and BC for food traceability are possible future work, while incorporating pluggable smart contracts could make traceability more flexible. The integration of these technologies has the potential to improve the efficiency of several processes, especially in food safety, and requires more in-depth research. In addition, there is a need to assess the utility of blockchain-based digital marketplaces for future food traceability and information security by creating and evaluating them, for example, in combination with blockchain platforms.

Author Contributions: Conceptualization, M.L. and C.S.; methodology, M.L. and T.L.; validation, M.L., L.X. and T.L.; formal analysis, S.L.; investigation, M.L.; resources, M.L.; data curation, M.L.; writing—original draft preparation, M.L.; writing—review and editing, M.L.; visualization, S.L.; supervision, C.S.; project administration, S.L.; funding acquisition, T.L. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by partly the National Natural Science Foundation of China under Grant 61871475, in part by the Special Project of Laboratory Construction of Guangzhou Innovation Platform Construction Plan under Grant 201905010006, Guangzhou Key Research and Development Project under Grant 202103000033, 201903010043, Guangdong Science and Technology Project under Grant 2020A1414050060, 2020B0202080002, Innovation Team Project of Universities in Guangdong Province under Grant 2021KCXTD019, Characteristic Innovation Project of Universities in Guangdong Province under Grant KA190578826, Guangdong Province Enterprise Science and Technology Commissioner Project under Grant GDKTP2021004400, Guangdong Science and Technology Planning Project under Grant 2016A020210122, Meizhou City S&T Planned Projects under Grant 2021A0305010, Rural Science and Technology Correspondent Project of Zengcheng District, Guangzhou City under Grant 2021B42121631, Educational Science Planning Project of Guangdong Province under Grant 2020GXJK102, and Grant 2018GXJK072, Guangdong Province Graduate Education Innovation Program Project under Grant 2022XSLT056, 2022JGXM115, Guangzhou Science and Technology Planning Project under Grant 201903010043, Guangdong Science and Technology Planning Project under Grant 2015A040405014, the National Natural Science Foundation of China under Grant 31871525.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

References

- Omarov, R.; Agarkov, A.; Rastovarov, E.; Shlykov, S. Modern methods for food safety. *Eng. Rural Dev. Proc.* 2017, 960–963. [CrossRef]
- Botcha, K.M.; Chakravarthy, V.V.; Anurag; Enhancing traceability in pharmaceutical supply chain using Internet of Things (IoT) and blockchain. In Proceedings of the 2019 IEEE International Conference on Intelligent Systems and Green Technology (ICISGT). Visakhapatnam, India, 29–30 June 2019; pp. 45–453.
- Nordhagen, S. Food safety perspectives and practices of consumers and vendors in Nigeria: A review. *Food Control* 2022, 134, 108693. [CrossRef]
- Gunders, D.; Bloom, J. Wasted: How America Is Losing up to 40 Percent of Its Food from Farm to Fork to Landfill. NRDC Issue Paper 2017, 1–26.
- Moldovan, M.-G.; Dabija, D.-C.; Pocol, C.B. Resources Management for a Resilient World: A Literature Review of Eastern European Countries with Focus on Household Behaviour and Trends Related to Food Waste. *Sustainability* 2022, 14, 7123. [CrossRef]
- 6. Liu, C.; Jiang, H.; Badulescu, D.; Bac, D.P. Achieving Zero Hunger Goal through Minimizing Waste in Food Supply Chain: Evidence from Asian Emerging Region. *Sustainability* **2022**, *14*, 5930. [CrossRef]
- Liu, W.; Wei, S.; Wang, S.; Lim, M.K.; Wang, Y. Problem identification model of agricultural precision management based on smart supply chains: An exploratory study from China. J. Clean. Prod. 2022, 352, 131622. [CrossRef]
- Biegańska, M. Intelligent packaging as a tool in supply chain traceability in the food sector. In *Current Trends in Quality Science–Consumer Behavior, Logistic, Product Management;* Instytut Technologii Eksploatacji: Radom, Poland, 2022; pp. 226–236. Available online: https://ue.poznan.pl/pl/uniwersytet,c13/instytut-nauk-o-jakosci,c13079/konferencje,c14716/icqmsc-2021,c14717/monographs-current-trends-in-quality-sciences-2021,a105563.html (accessed on 22 July 2022).
- Fan, Y.; Lin, X.; Liang, W.; Wang, J.; Tan, G.; Lei, X.; Jing, L. TraceChain: A blockchain-based scheme to protect data confidentiality and traceability. Softw. Pract. Exp. 2022, 52, 115–129. [CrossRef]

- 10. Marsden, T.; Banks, J.; Bristow, G. Food supply chain approaches: Exploring their role in rural development. *Sociol. Rural* **2000**, 40, 424–438. [CrossRef]
- 11. Chen, W.; Chen, C.-L.; Cao, Q.; Chiu, C.-H. Time course and epidemiological features of COVID-19 resurgence due to cold-chain food or packaging contamination. *Biomed. J.* 2022, *in press.* [CrossRef]
- Peng, X.; Zhang, X.; Wang, X.; Li, H.; Xu, J.; Zhao, Z. Multi-Chain Collaboration-Based Information Management and Control for the Rice Supply Chain. *Agriculture* 2022, 12, 689. [CrossRef]
- 13. Liu, S.; Mu, T.; Xu, S.; He, G. Research on cross-chain method based on distributed Digital Identity. In Proceedings of the 2022 4th International Conference on Blockchain Technology, New York, NY, USA, 25–27 March 2022; pp. 59–73.
- Wu, X.; Li, Y.; Li, J.; Sun, Y. Research on the traceability coding of grain and oil products based on the EAN-UCC coding system. In Proceedings of the 2022 IEEE 2nd International Conference on Power, Electronics and Computer Applications (ICPECA), Shenyang, China, 21–23 January 2022.
- 15. Prashar, D.; Jha, N.; Jha, S.; Lee, Y.; Joshi, G.P. Blockchain-based traceability and visibility for agricultural products: A decentralized way of ensuring food safety in india. *Sustainability* **2020**, *12*, 3497. [CrossRef]
- 16. Yu, Z.; Jung, D.; Park, S.; Hu, Y.; Huang, K.; Rasco, B.A.; Wang, S.; Ronholm, J.; Lu, X.; Chen, J. Smart traceability for food safety. *Crit. Rev. Food Sci. Nutr.* **2022**, *62*, 905–916. [CrossRef]
- 17. Qian, J.; Dai, B.; Wang, B.; Zha, Y.; Song, Q. Traceability in food processing: Problems, methods, and performance evaluations—A review. *Crit. Rev. Food Sci. Nutr.* 2022, *62*, 679–692. [CrossRef]
- Manning, L.; Brewer, S.; Craigon, P.J.; Frey, J.; Gutierrez, A.; Jacobs, N.; Kanza, S.; Munday, S.; Sacks, J.; Pearson, S. Artificial intelligence and ethics within the food sector: Developing a common language for technology adoption across the supply chain. *Trends Food Sci. Technol.* 2022, 125, 33–42. [CrossRef]
- 19. Wright, C.S. Bitcoin: A Peer-to-Peer Electronic Cash System. pp. 1–9. Available online: https://ssrn.com/abstract=3440802 (accessed on 22 July 2022). [CrossRef]
- 20. Tan, A.; Gligor, D.; Ngah, A. Applying blockchain for halal food traceability. Int. J. Logist. Res. Appl. 2020, 25, 947–964. [CrossRef]
- Anwar, M.R.; Apriani, D.; Adianita, I.R. Hash Algorithm In Verification of Certificate Data Integrity And Security. *Aptisi Trans. Technopreneurship* 2021, 3, 181–188. [CrossRef]
- 22. Zheng, W.; Zheng, Z.; Chen, X.; Dai, K.; Li, P.; Chen, R. Nutbaas: A blockchain-as-a-service platform. *IEEE Access* 2019, 7, 134422–134433. [CrossRef]
- Nair, P.R.; Dorai, D.R. Evaluation of Performance and Security of Proof of Work and Proof of Stake using Blockchain. In Proceedings of the 2021 Third International Conference on Intelligent Communication Technologies and Virtual Mobile Networks (ICICV), Tirunelveli, India, 4–6 February 2021; pp. 279–283.
- 24. Creydt, M.; Fischer, M. Blockchain and more-Algorithm driven food traceability. Food Control 2019, 105, 45–51. [CrossRef]
- Nasir, M.H.; Arshad, J.; Khan, M.M.; Fatima, M.; Salah, K.; Jayaraman, R. Scalable blockchains—A systematic review. *Future Gener. Comput. Syst.* 2022, 126, 136–162. [CrossRef]
- Kadadha, M.; Otrok, H.; Mizouni, R.; Singh, S.; Ouali, A. On-chain behavior prediction Machine Learning model for blockchainbased crowdsourcing. *Future Gener. Comput. Syst.* 2022, 136, 170–181. [CrossRef]
- Tanya, R.; Chelliah, B.J. A Comprehensive Study on Cybersecurity Challenges and Solutions in an IoT Framework. In *Towards a Wireless Connected World: Achievements and New Technologies*; Springer: Berlin/Heidelberg, Germany, 2022; pp. 105–137.
- Hallak, J.C.; Tacsir, A. Traceability systems as a differentiation tool in agri-food value chains: A framework for public policies in Latin America. J. Agribus. Dev. Emerg. Econ. 2022, ahead-of-print. [CrossRef]
- 29. Nandakumar, K.; Sumathi, P.; Sanjaykumar, S.; Radhakrishnan, P.; Vasanthi, N.; Sandhiya, S. Food Safety Transparency and Traceability System Based on ML Enabled Blockchain Using Design Thinking. J. Posit. Sch. Psychol. 2022, 6, 5414–5424.
- 30. Naik, G.; Suresh, D.N. Challenges of creating sustainable agri-retail supply chains. IIMB Manag. Rev. 2018, 30, 270–282. [CrossRef]
- Tsang, Y.P.; Choy, K.L.; Wu, C.H.; Ho, G.T.S.; Lam, H.Y. Blockchain-driven IoT for food traceability with an integrated consensus mechanism. *IEEE Access* 2019, 7, 129000–129017. [CrossRef]
- 32. Ashton, K. That 'internet of things' thing. RFID J. 2009, 22, 97–114.
- 33. Yin, H.; Cao, Y.; Marelli, B.; Zeng, X.; Mason, A.J.; Cao, C. Soil Sensors and Plant Wearables for Smart and Precision Agriculture. *Adv. Mater.* **2021**, *33*, e2007764. [CrossRef]
- 34. Urdampilleta, M.; Ayela, C.; Ducrot, P.-H.; Rosario-Amorin, D.; Mondal, A.; Rouzières, M.; Dechambenoit, P.; Mathonière, C.; Mathieu, F.; Dufour, I.; et al. Molecule-based microelectromechanical sensors. *Sci. Rep.* **2018**, *8*, 8016. [CrossRef]
- Sharmila, S.; Vijayarani, S. Machine to Machine (M2M), Radio-frequency Identification (RFID), and Software-Defined Networking (SDN): Facilitators of the Internet of Things. In *Artificial Intelligence-Based Internet of Things Systems*; Springer: Berlin/Heidelberg, Germany, 2022; pp. 219–242.
- El Matbouly, H.; Nikbakhtnasrabadi, F.; Dahiya, R. RFID Near-field Communication (NFC)-Based Sensing Technology in Food Quality Control. In *Biosensing and Micro-Nano Devices*; Springer: Singapore, 2022; pp. 219–241.
- Ye, H.; Lee, C.-J.; Wu, T.-Y.; Yang, X.-D.; Chen, B.-Y.; Liang, R.-H. Body-Centric NFC: Body-Centric Interaction with NFC Devices through Near-Field Enabled Clothing. In Proceedings of the Designing Interactive Systems Conference, Online, 13–17 June 2022; pp. 1626–1639.
- Tan, W.C.; Sidhu, M.S. Review of RFID and IoT integration in supply chain management. Oper. Res. Perspect. 2022, 9, 100229. [CrossRef]

- 39. Ariza, J.A.; Baez, H. Understanding the role of single-board computers in engineering and computer science education: A systematic literature review. *Comput. Appl. Eng. Educ.* **2022**, *30*, 304–329. [CrossRef]
- Aronson, J.K. Artificial Intelligence in Pharmacovigilance: An Introduction to Terms, Concepts, Applications, and Limitations. Drug Saf. 2022, 45, 407–418. [CrossRef]
- Ou, W.; Gao, W.; Li, Z.; Zhang, S.; Wang, Q. Application of keywords speech recognition in agricultural voice information system. In Proceedings of the 2010 Second International Conference on Computational Intelligence and Natural Computing, Wuhan, China, 13–14 September 2010; pp. 197–200.
- 42. Tran, N.K.; Albahra, S.; May, L.; Waldman, S.; Crabtree, S.; Bainbridge, S.; Rashidi, H. Evolving applications of artificial intelligence and machine learning in infectious diseases testing. *Clin. Chem.* **2022**, *68*, 125–133. [CrossRef] [PubMed]
- Tian, H.; Wang, T.; Liu, Y.; Qiao, X.; Li, Y. Computer vision technology in agricultural automation—A review. *Inf. Process. Agric.* 2020, 7, 1–19. [CrossRef]
- Vasconez, J.P.; Kantor, G.A.; Cheein, F.A.A. Human–robot interaction in agriculture: A survey and current challenges. *Biosyst.* Eng. 2019, 179, 35–48. [CrossRef]
- Dewi, T.; Risma, P.; Oktarina, Y.; Muslimin, S. Visual servoing design and control for agriculture robot; a review. In Proceedings of the 2018 International Conference on Electrical Engineering and Computer Science (ICECOS), Pangkal Pinang, Indonesia, 2–4 October 2018; pp. 57–62.
- 46. Debats, S.R.; Luo, D.; Estes, L.D.; Fuchs, T.J.; Caylor, K.K. A generalized computer vision approach to mapping crop fields in heterogeneous agricultural landscapes. *Remote Sens. Environ.* **2016**, *179*, 210–221. [CrossRef]
- 47. Ingram, J.; Maye, D.; Bailye, C.; Barnes, A.; Bear, C.; Bell, M.; Cutress, D.; Davies, L.; de Boon, A.; Dinnie, L.; et al. What are the priority research questions for digital agriculture? *Land Use Policy* **2022**, *114*, 105962. [CrossRef]
- Wang, X. Managing Land Carrying Capacity: Key to Achieving Sustainable Production Systems for Food Security. Land 2022, 11, 484. [CrossRef]
- Feng, L.; Zhang, H.; Lou, L.; Chen, Y. A blockchain-based collocation storage architecture for data security process platform of WSN. In Proceedings of the 2018 IEEE 22nd International Conference on Computer Supported Cooperative Work in Design (CSCWD), Nanjing, China, 9–11 May 2018; pp. 75–80.
- 50. Showail, A.; Tahir, R.; Zaffar, M.F.; Noor, M.H.; Al-Khatib, M. An Internet of Secure and Private Things: A Service-Oriented Architecture. *Comput. Secur.* 2022, 120, 102776. [CrossRef]
- 51. Hyperledger Fabric Documentation. 2022. Available online: https://hyperledgerfabric.readthedocs.io/en/release-2.2/ (accessed on 22 July 2022).
- 52. Hassan, M.U.; Rehmani, M.H.; Chen, J. Differential privacy techniques for cyber physical systems: A survey. *IEEE Commun. Surv. Tutor.* **2020**, *22*, 746–789. [CrossRef]
- Inan, A.; Kantarcioglu, M.; Ghinita, G.; Bertino, E. Private record matching using differential privacy. In Proceedings of the 13th International Conference on Extending Database Technology, Lausanne, Switzerland, 20–26 March 2010; pp. 123–134.
- 54. Mothukuri, V.; Parizi, R.M.; Pouriyeh, S.; Huang, Y.; Dehghantanha, A.; Srivastava, G. A survey on security and privacy of federated learning. *Futur. Gener. Comput. Syst.* 2021, 115, 619–640. [CrossRef]
- 55. Triastcyn, A.; Faltings, B. Federated generative privacy. IEEE Intell. Syst. 2020, 35, 50–57. [CrossRef]
- 56. Wu, Y.; Wang, X.; Susilo, W.; Yang, G.; Jiang, Z.L.; Yiu, S.-M.; Wang, H. Generic server-aided secure multi-party computation in cloud computing. *Comput. Stand. Interfaces* 2022, 79, 103552. [CrossRef]
- Luongo, M.; Pon, C. The Keep Network: A Privacy Layer for Public Blockchains. 2018. Available online: https://static.coinpaprika. com/storage/cdn/whitepapers/448413.pdf (accessed on 22 July 2022).
- 58. Mu, C.; Ding, T.; Sun, Y.; Huang, Y.; Li, F.; Siano, P. Energy Block-based Peer-to-Peer Contract Trading with Secure Multi-party Computation in Nanogrid. *IEEE Trans. Smart Grid* **2022**, *8*, 19410–19432. [CrossRef]
- Peng, S.; Cai, Z.; Liu, W.; Wang, W.; Li, G.; Sun, Y.; Zhu, L. Blockchain Data Secure Transmission Method Based on Homomorphic Encryption. *Comput. Intell. Neurosci.* 2022, 2022, 3406228. [CrossRef]
- Hellwig, D.P.; Huchzermeier, A. Distributed ledger technology and fully homomorphic encryption: Next-generation informationsharing for supply chain efficiency. In *Innovative Technology at the Interface of Finance and Operations*; Springer: Berlin/Heidelberg, Germany, 2022; pp. 31–49.
- Akram, M.; Sen, A. A case study Evaluation of Blockchain for digital identity verification and management in BFSI using Zero-Knowledge Proof. In Proceedings of the 2022 International Conference on Decision Aid Sciences and Applications (DASA), Chiangrai, Thailand, 23–25 March 2022; pp. 1295–1299.
- 62. Wen, X.-J.; Chen, Y.-Z.; Fan, X.-C.; Zhang, W.; Yi, Z.-Z.; Fang, J.-B. Blockchain consensus mechanism based on quantum zero-knowledge proof. *Opt. Laser Technol.* 2022, 147, 107693. [CrossRef]
- Yang, K.; Wang, X.; Non-Interactive Zero-Knowledge Proofs to Multiple Verifiers. Cryptology ePrint Archive. 2022, pp. 1–55. Available online: https://eprint.iacr.org/2022/063 (accessed on 22 July 2022).
- 64. Xu, J.; Gu, B.; Tian, G. Review of agricultural IoT technology. Artif. Intell. Agric. 2022, 6, 10–22. [CrossRef]
- Kuznetsov, A.; Poluyanenko, N.; Kiian, A.; Poliakova, O. Application of Bluetooth, Wi-Fi and GPS Technologies in the Means of Contact Tracking. In *Information Security Technologies in the Decentralized Distributed Networks*; Springer: Berlin/Heidelberg, Germany, 2022; pp. 33–69.

- 66. Kang, J.; Xiong, Z.; Niyato, D.; Ye, D.; Kim, D.I.; Zhao, J. Toward secure blockchain-enabled internet of vehicles: Optimizing consensus management using reputation and contract theory. *IEEE Trans. Veh. Technol.* **2019**, *68*, 2906–2920. [CrossRef]
- 67. Kuhn, M.; Funk, F.; Zhang, G.; Franke, J. Blockchain-based application for the traceability of complex assembly structures. *J. Manuf. Syst.* **2021**, *59*, 617–630. [CrossRef]
- Androulaki, E.; Barger, A.; Bortnikov, V.; Cachin, C.; Christidis, K.; De Caro, A.; Enyeart, D.; Ferris, C.; Laventman, G.; Manevich, Y.; et al. Hyperledger fabric: A distributed operating system for permissioned blockchains. In Proceedings of the Thirteenth EuroSys Conference, New York, NY, USA, 23–26 April 2018; pp. 1–15.
- 69. Zyskind, G.; Nathan, O. Decentralizing privacy: Using blockchain to protect personal data. In Proceedings of the Security and Privacy Workshops (SPW), San Jose, CA, USA, 21–22 May 2015; pp. 180–184.
- 70. Wang, L.; He, Y.; Wu, Z. Design of a Blockchain-Enabled Traceability System Framework for Food Supply Chains. *Foods* 2022, 11, 744. [CrossRef]
- Pateriya, R.K.; Sharma, S. Te evolution of RFID security and privacy: A research survey. In Proceedings of the 2011 International Conference on Communication Systems and Network Technologies, Jammu, India, 3–5 June 2011; pp. 115–119.
- Juels, A.; Rivest, R.L.; Szydlo, M. The blocker tag: Selective blocking of RFID tags for consumer privacy. In Proceedings of the 10th ACM conference on Computer and Communications Security, Singapore, 14 April–17 March 2015; pp. 103–111.
- Yuvaraj, N.; Raja, R.A.; Karthikeyan, T.; Kousik, N.V. Improved privacy preservation framework for cloud-based internet of things. In *Internet of Things*; CRC Press: Boca Raton, FL, USA, 2020; pp. 165–174.
- 74. Shrestha, R.; Kim, S. Integration of IoT with blockchain and homomorphic encryption: Challenging issues and opportunities. In *Advances in Computers*; Elsevier: Amsterdam, The Netherlands, 2019; Volume 115, pp. 293–331.
- Gervais, A.; Karame, G.O.; Wüst, K.; Glykantzis, V.; Ritzdorf, H.; Capkun, S. On the security and performance of proof of work blockchains. In Proceedings of the 2016 ACM SIGSAC Conference on Computer and Communications Security, Vienna, Austria, 24–28 October 2016; pp. 3–16.
- Mirabelli, G.; Solina, V. Blockchain and agricultural supply chains traceability: Research trends and future challenges. *Procedia* Manuf. 2020, 42, 414–421. [CrossRef]
- 77. Lavelli, V.; Beccalli, M.P. Cheese whey recycling in the perspective of the circular economy: Modeling processes and the supply chain to design the involvement of the small and medium enterprises. *Trends Food Sci. Technol.* **2022**, *126*, 86–98. [CrossRef]
- Syed, N.F.; Shah, S.W.; Trujillo-Rasua, R.; Doss, R. Traceability in supply chains: A Cyber security analysis. *Comput. Secur.* 2022, 112, 102536. [CrossRef]
- Sun, W. Application of Blockchain Technology in the Supply Chain Finance. In Proceedings of the IEEE International Conference on Cloud Computing and Big Data Analysis (ICCCBDA), Chengdu, China, 22–24 April 2022; pp. 205–209.
- Huang, S.; Wang, G.; Yan, Y.; Fang, X. Blockchain-based data management for digital twin of product. J. Manuf. Syst. 2020, 54, 361–371. [CrossRef]
- Tian, H.; Jian, Y.; Ge, X. Blockchain-based AMI framework for data security and privacy protection. *Sustain. Energy Grids Netw.* 2022, 32, 100807. [CrossRef]
- Zhou, P.; Wang, K.; Guo, L.; Gong, S.; Zheng, B. A Privacy-Preserving Distributed Contextual Federated Online Learning Framework with Big Data Support in Social Recommender Systems. *IEEE Trans. Knowl. Data Eng.* 2021, 33, 824–838. [CrossRef]
- Zhang, X.; Sun, P.; Xu, J.; Wang, X.; Yu, J.; Zhao, Z.; Dong, Y. Blockchain-based safety management system for the grain supply chain. *IEEE Access* 2020, *8*, 36398–36410. [CrossRef]
- Truex, S.; Baracaldo, N.; Anwar, A.; Steinke, T.; Ludwig, H.; Zhang, R.; Zhou, Y. A hybrid approach to privacy-preserving federated learning. In Proceedings of the 12th ACM Workshop on Artificial Intelligence and Security, London, UK, 15 November 2019; ACM: New York, NY, USA, 2019; pp. 1–11. [CrossRef]
- Hua, J.; Wang, X.; Kang, M.; Wang, H.; Wang, F.-Y. Blockchain Based Provenance for Agricultural Products: A Distributed Platform with Duplicated and Shared Bookkeeping. In Proceedings of the 2018 IEEE Intelligent Vehicles Symposium (IV), Changshu, China, 26–30 June 2018; pp. 97–101. [CrossRef]
- 86. Casado-Vara, R.; Prieto, J.; De la Prieta, F.; Corchado, J.M. How blockchain improves the supply chain: Case study alimentary supply chain. *Procedia Comput. Sci.* 2018, 134, 393–398. [CrossRef]
- Wortmann Felix, C.M.; Andreas, B.; Dominik, B.; Elgar, F. Blockchain for the IoT: Privacy-Preserving Protection of Sensor Data. J. Assoc. Inf. Syst. 2019, 20, 1274–1309.
- Pincheira, C.; Miguel; Salek, A.; Muhammad; Vecchio, M.; Giaffreda, R. Blockchain-based traceability in Agri-Food supply chain management: A practical implementation. In Proceedings of the 2018 IoT Vertical and Topical Summit on Agriculture—Tuscany (IOT Tuscany), Tuscany, Italy, 8–9 May 2018; 9 May 2018; pp. 1–4.
- Kim, M.; Hilton, B.; Burks, Z.; Reyes, J. Integrating Blockchain, Smart Contract-Tokens, and IoT to Design a Food Traceability Solution. In Proceedings of the 2018 IEEE 9th Annual Information Technology, Electronics and Mobile Communication Conference (IEMCON), Vancouver, BC, Canada, 1–3 November 2018; pp. 335–340.
- 90. Baralla, G.; Ibba, S.; Marchesi, M.; Tonelli, R.; Missineo, S. A Blockchain Based System to Ensure Transparency and Reliability in Food Supply Chain. In *European Conference on Parallel Processing*; Springer: Cham, Switzerland, 2018; pp. 379–391.
- 91. Peng, X.; Zhang, X.; Wang, X.; Li, H.; Xu, J.; Zhao, Z.; Wang, Y. Research on the Cross-Chain Model of Rice Supply Chain Supervision Based on Parallel Blockchain and Smart Contracts. *Foods* **2022**, *11*, 1269. [CrossRef]

- Tse, D.; Zhang, B.; Yang, Y.; Cheng, C.; Mu, H. Blockchain application in food supply information security. In Proceedings of the 2017 IEEE International Conference on Industrial Engineering and Engineering Management (IEEM), Singapore, 10–13 December 2017; pp. 1357–1361.
- 93. Shaik, C. Preventing counterfeit products using cryptography, qr code and webservice. *Comput. Sci. Eng. Int. J.* (*CSEIJ*) 2021, 11, 1–11. [CrossRef]
- Carter, C.A.; Zhong, F.; Zhu, J. Advances in Chinese agriculture and its global implications. *Appl. Econ. Perspect. Policy* 2012, 34, 1–36. [CrossRef]
- 95. Kevorchian, C.; Gavrilescu, C.; Hurduzeu, G. A Peer-to-Peer (p2p) Agricultural Insurance Approach Based on Smart Contracts in Blockchain Ethereum. *Agric. Econ. Rural Dev. New Ser.* **2020**, *1*, 29–45.
- 96. Singh, M. Can Blockchain Technology Support Agricultural Sustainability? Authorea Preprints 2022, 1-44.
- 97. Kim, H.M.; Laskowski, M. Agriculture on the blockchain: Sustainable solutions for food, farmers, and financing. In *Supply Chain Revolution, Barrow Books*; AMACOM: New York, NY, USA, 2018.
- 98. Shen, M.; Tang, X.; Zhu, L.; Du, X.; Guizani, M. Privacy-preserving support vector machine training over blockchain-based encrypted IoT data in smart cities. *IEEE Internet Things J.* **2019**, *6*, 7702–7712. [CrossRef]
- 99. Xia, Q.; Sifah, E.B.; Smahi, A.; Amofa, S.; Zhang, X. BBDS: Blockchain-based data sharing for electronic medical records in cloud environments. *Information* **2017**, *8*, 44. [CrossRef]
- 100. Hassan, M.U.; Rehmani, M.H.; Chen, J. Privacy preservation in blockchain based IoT systems: Integration issues, prospects, challenges, and future research directions. *Future Gener. Comput. Syst.* 2019, 97, 512–529. [CrossRef]
- Rahman, M.A.; Hossain, M.S.; Islam, M.S.; Alrajeh, N.A.; Muhammad, G. Secure and provenance enhanced internet of health things framework: A blockchain managed federated learning approach. *IEEE Access* 2020, *8*, 205071–205087. [CrossRef]
- 102. George, R.V.; Harsh, H.O.; Ray, P.; Babu, A.K. Food quality traceability prototype for restaurants using blockchain and food quality data index. *J. Clean. Prod.* 2019, 240, 118021. [CrossRef]
- Zheng, M.; Zhang, S.; Zhang, Y.; Hu, B. Construct Food Safety Traceability System for People's Health Under the Internet of Things and Big Data. *IEEE Access* 2021, 9, 70571–70583. [CrossRef]
- Masudin, I.; Ramadhani, A.; Restuputri, D.P.; Amallynda, I. The effect of traceability system and managerial initiative on indonesian food cold chain performance: A COVID-19 pandemic perspective. *Glob. J. Flex. Syst. Manag.* 2021, 22, 331–356. [CrossRef]
- Katsikouli, P.; Wilde, A.S.; Dragoni, N.; Høgh-Jensen, H. On the benefits and challenges of blockchains for managing food supply chains. J. Sci. Food Agric. 2021, 101, 2175–2181. [CrossRef]
- 106. Köhler, S.; Pizzol, M. Technology assessment of blockchain-based technologies in the food supply chain. J. Clean. Prod. 2020, 269, 122193. [CrossRef]
- 107. Cozzolino, D. Recent trends on the use of infrared spectroscopy to trace and authenticate natural and agricultural food products. *Appl. Spectrosc. Rev.* **2012**, *47*, 518–530. [CrossRef]
- 108. Lyu, D.; Zhou, L.; Ma, X. Income-increasing effect of e-commerce of agricultural products based on IT absorptive theory. *Wirel. Netw.* **2022**, *28*, 2337–2344. [CrossRef]
- Chao, S. Construction Model of E-Commerce Agricultural Product Online Marketing System Based on Blockchain and Improved Genetic Algorithm. *Secur. Commun. Netw.* 2022, 2022, 4055698. [CrossRef]
- Demestichas, K.; Peppes, N.; Alexakis, T.; Adamopoulou, E. Blockchain in agriculture traceability systems: A review. *Appl. Sci.* 2020, 10, 4113. [CrossRef]
- Dabholkar, A.; Saraswat, V. Ripping the fabric: Attacks and mitigations on hyperledger fabric. In *Applications and Techniques in Information Security, Proceedings of the 10th International Conference, ATIS 2019, Thanjavur, India, 22–24 November 2019; Springer: Singapore, 2019; pp. 300–311.*
- 112. Tang, C.; Ma, Y.; Yu, X. Design and Implementation of Port Video Terminals Security Access Authentication System Using Blockchain Technology. In *E3S Web of Conferences*; EDP Sciences: Les Ulis, France, 2021; p. 03086.
- 113. Davda, Y.; Sunitha, R.; Honnavalli, P.B. Design of Hash Algorithm for Blockchain Security; EasyChair: Bangalore, India, 2022; pp. 1–8.
- Devika, K.N.; Bhakthavatchalu, R. Parameterizable FPGA implementation of SHA-256 using blockchain concept. In Proceedings of the 2019 International Conference on Communication and Signal Processing (ICCSP), Chennai, India, 4–6 April 2019; pp. 370–374.
- Sankar, L.S.; Sindhu, M.; Sethumadhavan, M. Survey of consensus protocols on blockchain applications. In Proceedings of the 2017 4th International Conference on Advanced Computing and Communication Systems (ICACCS), Coimbatore, India, 6–7 January 2017; pp. 1–5.
- 116. Bouraga, S. A taxonomy of blockchain consensus protocols: A survey and classification framework. *Expert Syst. Appl.* **2021**, *168*, 114384. [CrossRef]
- 117. Chen, D.; Orekondy, T.; Fritz, M. Gs-wgan: A gradient-sanitized approach for learning differentially private generators. *arXiv* **2020**, arXiv:2006.08265.2020.
- 118. Lan, S.; Hong, J.; Chen, J.; Cai, J.; Wang, Y. Equation Chapter 1 Section 1 Differentially Private High-Dimensional Binary Data Publication via Adaptive Bayesian Network. *Wirel. Commun. Mob. Comput.* **2021**, 2021, 8693978. [CrossRef]
- Wang, J.; Ling, X.; Le, Y.; Huang, Y.; You, X. Blockchain-enabled wireless communications: A new paradigm towards 6G. *Natl. Sci. Rev.* 2021, *8*, nwab069. [CrossRef]

- 120. Akbar, N.A.; Muneer, A.; ElHakim, N.; Fati, S.M. Distributed Hybrid Double-Spending Attack Prevention Mechanism for Proof-of-Work and Proof-of-Stake Blockchain Consensuses. *Future Internet* **2021**, *13*, 285. [CrossRef]
- 121. Gerrits, L.; Samuel, C.N.; Kromes, R.; Verdier, F.; Glock, S.; Guitton-Ouhamou, P. Experimental Scalability Study of Consortium Blockchains with BFT Consensus for IoT Automotive Use Case. In Proceedings of the 19th ACM Conference on Embedded Networked Sensor Systems, Coimbra, Portugal, 15–17 November 2021; pp. 492–498. [CrossRef]
- 122. Pandey, M.; Agarwal, R.; Shukla, S.K.; Verma, N.K. Reputation-based PoS for the Restriction of Illicit Activities on Blockchain: Algorand Usecase. *arXiv* 2021, arXiv:2112.11024. [CrossRef]
- 123. Zhao, H.; Deng, S.; Liu, Z.; Xiang, Z.; Yin, J.; Dustdar, S.; Zomaya, A. Dpos: Decentralized, privacy-preserving, and low-complexity online slicing for multi-tenant networks. *IEEE Trans. Mob. Comput.* **2021**. [CrossRef]
- 124. Yasir, A.I.; Kusuma, G.P. Deep Deterministic Policy Gradients for Optimizing Simulated PoA Blockchain Networks Based on Healthcare Data Characteristics. *Adv. Sci. Technol. Eng. Syst. J.* **2021**, *6*, 757–764. [CrossRef]
- 125. Yuan, R.; Yu-Bin, X.; Chen, H.-B.; Bin-Yu, Z.; Xie, J. Shadoweth: Private smart contract on public blockchain. J. Comput. Sci. Technol. 2018, 33, 542–556. [CrossRef]
- 126. Kalodner, H.; Goldfeder, S.; Chen, X.; Weinberg, S.M.; Felten, E.W. Arbitrum: Scalable, Private Smart Contracts; USENIX Association: Baltimore, MD, USA, 2018; pp. 1353–1370. Available online: https://www.usenix.org/conference/usenixsecurity18/presentation/kalodner (accessed on 22 July 2022).
- 127. Sánchez, D.C. Raziel: Private and verifiable smart contracts on blockchains. arXiv 2018, arXiv:1807.09484.
- Atzei, N.; Bartoletti, M.; Cimoli, T. Atzei, N.; Bartoletti, M.; Cimoli, T. A survey of attacks on ethereum smart contracts (sok). In International Conference on Principles of Security and Trust; Springer: Berlin/Heidelberg, Germany, 2017; Bartoletti, M.; Volume 10204, pp. 164–186.
- 129. Wen, Q.; Gao, Y.; Chen, Z.; Wu, D. A blockchain-based data sharing scheme in the supply chain by IIoT. In Proceedings of the 2019 IEEE International Conference on Industrial Cyber Physical Systems (ICPS), Taipei, Taiwan, 6–9 May 2019; pp. 695–700.
- 130. Lin, Q.; Wang, H.; Pei, X.; Wang, J. Food safety traceability system based on blockchain and EPCIS. *IEEE Access* 2019, 7, 20698–20707. [CrossRef]
- Lin, C.; He, D.; Huang, X.; Xie, X.; Choo, K.-K.R. Ppchain: A privacy-preserving permissioned blockchain architecture for cryptocurrency and other regulated applications. *IEEE Syst. J.* 2020, 15, 4367–4378. [CrossRef]
- Pooja, S.; Mundada, M.R. Analysis of agricultural supply chain management for traceability of food products using blockchainethereum technology. In Proceedings of the 2020 IEEE International Conference on Distributed Computing, VLSI, Udupi, India, 30–31 October 2020, Electrical Circuits and Robotics (DISCOVER); pp. 127–132.
- 133. Majdalawieh, M.; Nizamuddin, N.; Alaraj, M.; Khan, S.; Bani-Hani, A. Blockchain-based solution for Secure and Transparent Food Supply Chain Network. *Peer-Peer Netw. Appl.* **2021**, *14*, 3831–3850. [CrossRef]
- 134. Koval, M. Blockchain in Dairy Supply Chains: A Literature Review. Master's Thesis, University of Twente, Enschede, The Netherlands, 2022.
- Guo, J.; Li, C.; Luo, Y. Blockchain-assisted caching optimization and data storage methods in edge environment. *J. Supercomput.* 2022, 1–33. [CrossRef]
- Daulatrao, B.C.; Nandkumar, D.P.; Atmaram, J.G.; Dattatray, P.Y.; Mohanrao, S.S. Agri Food Supply Chain Using Ethereum Smart Contract. Int. J. Mod. Dev. Eng. Sci. 2022, 1, 11–16.
- Marchesi, L.; Mannaro, K.; Marchesi, M.; Tonelli, R. Automatic Generation of Ethereum-Based Smart Contracts for Agri-Food Traceability System. *IEEE Access* 2022, 10, 50363–50383. [CrossRef]
- Hayati, H.; Nugraha, I.G.B.B. Blockchain based traceability system in food supply chain. In Proceedings of the 2018 International Seminar on Research of Information Technology and Intelligent Systems (ISRITI), Yogyakarta, Indonesia, 21–22 November 2018; pp. 120–125.
- Mohan, T. Improve Food Supply Chain Traceability Using Blockchain. Master's Thesis, The Pennsylvania State University, State College, PA, USA, 2018.
- Baralla, G.; Pinna, A.; Corrias, G. Ensure traceability in European food supply chain by using a blockchain system. In Proceedings of the 2019 IEEE/ACM 2nd International Workshop on Emerging Trends in Software Engineering for Blockchain (WETSEB), Montreal, QC, Canada, 27–27 May 2019; pp. 40–47.
- 141. Chan, K.Y.; Abdullah, J.; Khan, A.S. A framework for traceable and transparent supply chain management for agri-food sector in malaysia using blockchain technology. *Int. J. Adv. Comput. Sci. Appl.* **2019**, *10*, 149–156. [CrossRef]
- 142. Bumblauskas, D.; Mann, A.; Dugan, B.; Rittmer, J. A blockchain use case in food distribution: Do you know where your food has been? *Int. J. Inf. Manag.* 2020, 52, 102008. [CrossRef]
- Pincheira, M.; Vecchio, M.; Giaffreda, R. Benchmarking Constrained IoT Devices in Blockchain-Based Agri-Food Traceability Applications. Int. Congress Blockchain Appl. 2021, 320, 212–221. [CrossRef]
- Wisessing, K.; Vichaidis, N. IoT Based Cold Chain Logistics with Blockchain for Food Monitoring Application. In Proceedings of the 2022 7th International Conference on Business and Industrial Research (ICBIR), Bangkok, Thailand, 19–20 May 2022; pp. 359–363.
- 145. Mohit, M.; Kaur, S.; Singh, M. Design and implementation of transaction privacy by virtue of ownership and traceability in blockchain based supply chain. *Cluster Comput.* **2022**, *25*, 2223–2240. [CrossRef]

- 146. Wan, J.; Li, J.; Imran, M.; Li, D. A blockchain-based solution for enhancing security and privacy in smart factory. *IEEE Trans. Ind. Inform.* **2019**, *15*, 3652–3660. [CrossRef]
- 147. Sahai, S.; Singh, N.; Dayama, P. Enabling Privacy and Traceability in Supply Chains using Blockchain and Zero Knowledge Proofs. In Proceedings of the 2020 IEEE International Conference on Blockchain (Blockchain), Rhodes, Greece, 2–6 November 2020; pp. 134–143.
- 148. Balasubramanian, S.; Akila, I.S. Blockchain implementation for agricultural food supply Chain using hyperledger fabric. *J. Intell. Fuzzy Syst.* 2022; *pre-press.* [CrossRef]
- 149. Balamurugan, S.; Ayyasamy, A.; Joseph, K.S. IoT-Blockchain driven traceability techniques for improved safety measures in food supply chain. *Int. J. Inf. Tecnol.* 2022, 14, 1087–1098. [CrossRef]
- 150. Lu, Y.; Li, P.; Xu, H. A Food anti-counterfeiting traceability system based on Blockchain and Internet of Things. *Procedia Comput. Sci.* **2022**, *199*, 629–636. [CrossRef]
- 151. Gallo, P.; Daidone, F.; Sgroi, F.; Avantaggiato, M. AgriChain: Blockchain Syntactic and Semantic Validation for Reducing Information Asymmetry in Agri-Food. In Proceedings of the 2022: 4th Distributed Ledger Technology Workshop, Rome, Italy, 20 June 2022; pp. 1–3166. Available online: http://ceur-ws.org/Vol-3166/paper08.pdf (accessed on 22 July 2022).
- 152. Ravi, D.; Ramachandran, S.; Vignesh, R.; Falmari, V.R.; Brindha, M. Privacy preserving transparent supply chain management through Hyperledger Fabric. *Blockchain Res. Appl.* **2022**, *3*, 100072. [CrossRef]
- 153. Zhang, Y.; Zhu, D.; Ren, X.; Shen, Y.; Cao, X.; Liu, H.; Li, J. Quality changes and shelf-life prediction model of postharvest apples using partial least squares and artificial neural network analysis. *Food Chem.* **2022**, *394*, 133526. [CrossRef]
- Himthani, P.; Dubey, G.P.; Sharma, B.M.; Taneja, A. Big Data Privacy and Challenges for Machine Learning. In Proceedings of the 2020 Fourth International Conference on I-SMAC (IoT in Social, Mobile, Analytics and Cloud) (I-SMAC), Palladam, India, 7–9 October 2020, pp. 707–713.
- 155. Onal, A.C.; Berat Sezer, O.; Ozbayoglu, M.; Dogdu, E. Weather data analysis and sensor fault detection using an extended IoT framework with semantics, big data, and machine learning. In Proceedings of the 2017 IEEE International Conference on Big Data (Big Data), Boston, MA, USA, 11–14 December 2017; pp. 2037–2046.
- 156. Liu, B.; Ding, M.; Shaham, S.; Rahayu, W.; Farokhi, F.; Lin, Z. When machine learning meets privacy: A survey and outlook. *ACM Comput. Surv.* **2021**, *54*, 1–36. [CrossRef]
- 157. Ryffel, T.; Trask, A.; Dahl, M.; Wagner, B.; Mancuso, J.; Rueckert, D.; Passerat-Palmbach, J. A generic framework for privacy preserving deep learning. *arXiv* 2018, arXiv:1811.04017.
- 158. Hang, L.; Ullah, I.; Kim, D.-H. A secure fish farm platform based on blockchain for agriculture data integrity. *Comput. Electron. Agric.* **2020**, *170*, 105251. [CrossRef]
- 159. Ajagbe, S.A.; Awotunde, J.B.; Adesina, A.O.; Achimugu, P.; Kumar, T.A. Internet of Medical Things (IoMT): Applications, Challenges, and Prospects in a Data-Driven Technology. *Intell. Healthc.* **2022**, 299–319. [CrossRef]
- 160. Takatsuji, T.; Watanabe, H.; Yamashita, Y. Blockchain technology to visualize the metrological traceability. *Precis. Eng.* **2019**, *58*, 1–6. [CrossRef]
- Kshetri, N. The Economics of Blockchain-Based Supply Chain Traceability in Developing Countries. *Computer* 2021, 54, 98–103. [CrossRef]
- Wan, L.; Eyers, D.; Zhang, H. Evaluating the impact of network latency on the safety of blockchain transactions. In Proceedings of the 2019 IEEE International Conference on Blockchain (Blockchain), Atlanta, GA, USA, 14–17 July 2019; pp. 194–201.
- 163. Zhao, C.; Zhang, S.; Wang, T.; Liew, S.C. Bodyless Block Propagation: TPS Fully Scalable Blockchain with Pre-Validation. *arXiv* **2022**, arXiv:2204.08769.
- 164. Sezer, B.B.; Topal, S.; Nuriyev, U. TPPSUPPLY: A traceable and privacy-preserving blockchain system architecture for the supply chain. *J. Inf. Secur. Appl.* **2022**, *66*, 103116. [CrossRef]
- 165. Li, H.; Han, D.; Tang, M. A privacy-preserving storage scheme for logistics data with assistance of blockchain. *IEEE Internet Things J.* **2021**, *9*, 4704–4720. [CrossRef]
- 166. Islam, M.; Rehmani, M.H.; Chen, J. Differential Privacy-based Permissioned Blockchain for Private Data Sharing in Industrial IoT. *arXiv* 2021, arXiv:2102.09857.
- 167. Liang, W.; Ji, N. Privacy challenges of IoT-based blockchain: A systematic review. Clust. Comput. 2021, 25, 2203–2221. [CrossRef]
- Wu, G.; Wang, S.; Ning, Z.; Zhu, B. Privacy-Preserved EMR Information Publishing and Sharing: A Blockchain-Enabled Smart Healthcare System. *IEEE J. Biomed. Health Inform.* 2021, 26, 1917–1927. [CrossRef]
- 169. Zhou, P.; Kortoçi, P.; Yau, Y.-P.; Finley, B.; Wang, X.; Braud, T.; Lee, L.-H.; Tarkoma, S.; Kangasharju, J.; Hui, P. AICP: Augmented Informative Cooperative Perception. *IEEE Trans. Intell. Transp. Syst.* **2022**, 1–14. [CrossRef]
- 170. Pang, G. The AI Chip Race. IEEE Intell. Syst. 2022, 37, 111–112. [CrossRef]
- 171. Ahmed, W.A.; MacCarthy, B.L. Blockchain technology in the supply chain: Learning from emerging ecosystems and industry consortia. In *Handbook on Digital Business Ecosystems*; Edward Elgar Publishing: Cheltenham, UK, 2022; pp. 367–386.
- 172. Salim, A.M.; Dabous, S.A. A review of critical success factors for solar home system implementation in public housing. *Int. J. Energy Sect. Manag.* 2022, *ahead-of-print.* [CrossRef]
- 173. Cai, L.; Sun, Y.; Zheng, Z.; Xiao, J.; Qiu, W. Blockchain in China. Commun. ACM 2021, 64, 88–93. [CrossRef]
- 174. Coad, A.; Harasztosi, P.; Pál, R.; Teruel, M.; others. Policy instruments for high-growth enterprises. *Quest. Entrep. State* 2022, 53, 273.

- 175. Benyam, A.A.; Soma, T.; Fraser, E. Digital agricultural technologies for food loss and waste prevention and reduction: Global trends, adoption opportunities and barriers. *J. Clean. Prod.* **2021**, *323*, 129099. [CrossRef]
- 176. Marin, M.-P.; Marin, I.; Vidu, L. Learning about the reduction of food waste using blockchain technology. *arXiv* 2021, arXiv:2101.02026.
- 177. Trollman, H.; Garcia-Garcia, G.; Jagtap, S.; Trollman, F. Blockchain for Ecologically Embedded Coffee Supply Chains. *Logistics* **2022**, *6*, 43. [CrossRef]
- 178. Gopalakrishnan, P.K.; Hall, J.; Behdad, S. Cost analysis and optimization of Blockchain-based solid waste management traceability system. *Waste Manag.* 2021, 120, 594–607. [CrossRef]
- 179. Sunny, J.; Undralla, N.; Pillai, V.M. Supply chain transparency through blockchain-based traceability: An overview with demonstration. *Comput. Ind. Eng.* 2020, 150, 106895. [CrossRef]
- Ferlay, J.; Colombet, M.; Soerjomataram, I.; Mathers, C.; Parkin, D.M.; Piñeros, M.; Znaor, A.; Bray, F. Estimating the global cancer incidence and mortality in 2018: GLOBOCAN sources and methods. *Int. J. Cancer* 2019, 144, 1941–1953. [CrossRef]
- 181. Garcia-Torres, S.; Albareda, L.; Rey-Garcia, M.; Seuring, S. Traceability for sustainability–literature review and conceptual framework. *Supply Chain. Manag. Int. J.* **2019**, *24*, 85–106. [CrossRef]
- 182. Gomiero, T.; Pimentel, D.; Paoletti, M.G. Is there a need for a more sustainable agriculture? *Crit. Rev. Plant Sci.* **2011**, *30*, 6–23. [CrossRef]
- Kaur, P.; Parashar, A. A systematic literature review of blockchain technology for smart villages. *Arch. Comput. Methods Eng.* 2021, 29, 2417–2468. [CrossRef] [PubMed]
- Kamilaris, A.; Fonts, A.; Prenafeta-Boldu, F.X. The rise of blockchain technology in agriculture and food supply chains. *Trends Food Sci. Technol.* 2019, *91*, 640–652. [CrossRef]
- 185. Funk, E.; Riddell, J.; Ankel, F.; Cabrera, D. Blockchain technology: A data framework to improve validity, trust, and accountability of information exchange in health professions education. *Acad. Med.* **2018**, *93*, 1791–1794. [CrossRef] [PubMed]
- Singh, S.; Sharma, P.K.; Yoon, B.; Shojafar, M.; Cho, G.H.; Ra, I.-H. Convergence of blockchain and artificial intelligence in IoT network for the sustainable smart city. *Sustain. Cities Soc.* 2020, 63, 102364. [CrossRef]
- 187. Blanco, C.; Wall, M.M.; Olfson, M. Data needs and models for the opioid epidemic. Mol. Psychiatry 2021, 27, 787–792. [CrossRef]
- Goldfeder, S.; Kalodner, H.; Reisman, D.; Narayanan, A. When the cookie meets the blockchain: Privacy risks of web payments via cryptocurrencies. arXiv 2017, arXiv:1708.04748. [CrossRef]